

THE UNIVERSITY of EDINBURGH

Edinburgh Research Explorer

The determinants of R&D smoothing with asset sales

Citation for published version:

Liu, D, Li, Z, He, H & Hou, W 2021, 'The determinants of R&D smoothing with asset sales: Evidence from R&D-intensive firms in China', International Review of Economics and Finance, vol. 75, pp. 76-93. https://doi.org/10.1016/j.iref.2021.03.013

Digital Object Identifier (DOI):

10.1016/j.iref.2021.03.013

Link:

Link to publication record in Edinburgh Research Explorer

Document Version: Peer reviewed version

Published In: International Review of Economics and Finance

General rights

Copyright for the publications made accessible via the Edinburgh Research Explorer is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy The University of Edinburgh has made every reasonable effort to ensure that Edinburgh Research Explorer content complies with UK legislation. If you believe that the public display of this file breaches copyright please contact openaccess@ed.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.



Journal Pre-proof

The determinants of R&D smoothing with asset sales: Evidence from R&D-intensive firms in China

Duan Liu, Zhiyuan Li, Hongbo He, Wenxuan Hou

PII: S1059-0560(21)00053-8

DOI: https://doi.org/10.1016/j.iref.2021.03.013

Reference: REVECO 2182

- To appear in: International Review of Economics and Finance
- Received Date: 30 December 2019
- Revised Date: 19 February 2021
- Accepted Date: 22 March 2021

Please cite this article as: Liu D., Li Z., He H. & Hou W., The determinants of R&D smoothing with asset sales: Evidence from R&D-intensive firms in China, *International Review of Economics and Finance* (2021), doi: https://doi.org/10.1016/j.iref.2021.03.013.

This is a PDF file of an article that has undergone enhancements after acceptance, such as the addition of a cover page and metadata, and formatting for readability, but it is not yet the definitive version of record. This version will undergo additional copyediting, typesetting and review before it is published in its final form, but we are providing this version to give early visibility of the article. Please note that, during the production process, errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

© 2021 Published by Elsevier Inc.



The determinants of R&D smoothing with asset sales: Evidence

from R&D-intensive firms in China

Duan Liu^a, Zhiyuan Li^a, Hongbo He^a* and Wenxuan Hou^b

^aDuan Liu (<u>perfect_done@163.com</u>), Zhiyuan Li (<u>jyyuan992@163.com</u>), and Hongbo He (*corresponding author, <u>hongbo_he@hnu.edu.cn</u>) are associate professor, PhD student, and associate professor, respectively, Business School, Hunan University, China.

^bWenxuan Hou (<u>wenxuan.hou@ed.ac.uk</u>) is Chair Professor in Corporate Finance, Business School, University of Edinburgh, UK.

Abstract

Research and development (R&D)-intensive firms have strong incentives to maintain a smooth path for their R&D investments; otherwise, they will incur high adjustment costs. Examining data covering 2009 to 2016, we find that Chinese R&D-intensive firms, especially those with high innovation efficiency, tend to sell operating and financial assets to protect their value-enhancing R&D investments. However, financial constraints have adverse impacts on R&D smoothing with asset sales unless the firms have high innovation efficiency. The results suggest that innovation efficiency offers R&D-intensive firms, even financially constrained ones, a strong motivation to covert asset sales proceeds into R&D inputs, as the proceeds from asset sales provide a less-costly substitute for external financing. Given the importance of R&D for economic growth and the limited external financing opportunities in emerging capital markets like China, our findings reveal new insights regarding R&D financing.

JEL classification: O32; C23; G34

Keywords: R&D financing; asset sale; innovation efficiency; financial constraint

Acknowledgments

The authors acknowledge financial support from the National Natural Science Foundation of China (71972066, 71573077, 71790593, 71521061, and 71673033), the Humanities and Social Science Foundation of the Ministry of Education of China (17YJA630056, 16YJA790058), and the Philosophy and Social Science Foundation of Hunan Province (18YBA085). All errors and omissions are the sole responsibility of the authors.

Declarations of interest: None

The determinants of R&D smoothing with assets sales: Evidence from R&D-intensive firms in China

Abstract

Research and development (R&D)-intensive firms have strong incentives to maintain a smooth path for their R&D investments; otherwise, they will incur high adjustment costs. Examining data covering 2009 to 2016, we find that Chinese R&D-intensive firms, especially those with high innovation efficiency, tend to sell operating and financial assets to protect their value-enhancing R&D investments. However, financial constraints have adverse impacts on R&D smoothing with asset sales unless the firms have high innovation efficiency. The results suggest that innovation efficiency offers R&D-intensive firms, even financially constrained ones, a strong motivation to covert asset sales proceeds into R&D inputs, as the proceeds from asset sales provide a less-costly substitute for external financing. Given the importance of R&D for economic growth and the limited external financing opportunities in emerging capital markets like China, our findings reveal new insights regarding R&D financing.

JEL classification: O32; C23; G34

Keywords: R&D financing; asset sale; innovation efficiency; financial constraint

1. Introduction

It is widely argued in the literature that research and development (R&D)-driven innovation is essential for productivity, economic growth and job creations (Solow, 1957; Grossman and Helpman, 1991; Coad, 2019). However, discontinued (or suspended) R&D investments are generally subject to a variety of adjustment costs, such as firing or rehiring costs for highly-trained employees, sunk costs, and time-compression diseconomies (Borisova and Brown, 2013; Brown and Petersen, 2015; Kang et al., 2017). It is thus strategically important for R&D-intensive firms to maintain a smooth path for their R&D investments.

However, R&D activity is susceptible to financial constraints owing to high uncertainty, asymmetric information, and limited collateral value (David et al., 2000; Hall, 2002; Brown et al., 2012). On the one hand, financially constrained firms are more likely to reject or scale down their innovative projects (Campello et al., 2010; Zhang, 2015), as financial constraints are generally believed to have negative effects on firms' innovation activities (Howell, 2016; Pellegrinoa et al., 2017).¹ As pointed out by Zhang(2015), for example, that R&D-intensive firms are more inclined to discontinue or suspend their R&D projects when facing severe financial constraints.

On the other hand, R&D-intensive firms may choose to protect their R&D investments through financing asset sales if they are financially constrained.² The proceeds from asset sales provide a less-costly substitute, especially when alternative sources of financing are either unavailable or too expensive (Lang et al., 1995; Shleifer and Vishny, 1992). In other words, financially constrained firms may choose to transfer cash inflows from financing asset sales to protect their value-enhancing R&D investment, which would otherwise be lost owing to high financing costs (Borisova and Brown, 2013; Brown and Petersen, 2015).

¹ Several studies, including Bhagat and Welch(1995), Stephen et al.(2003), Hoegl et al.(2008), and Pellegrinoa and Savona(2017), find that the relationship between financial constraints and innovation activities depends on market conditions.

² Arnold et al. (2018) define assets sale for financing as financing assets sale, while Brown and Petersen (2015) point out that asset sales provide a way of raising fund to finance new investment opportunities or meet liquidity needs. In this study, financing assets sale includes both operating assets sale and financial assets sale. According to Chinese Accounting Standards (CASs), operating assets are assets such as property and equipment that a company uses to produce its goods and services, rather than the goods or services that it sells. By contrast, financial assets mainly include tradable financial assets (short-term financial assets), held-to-maturity investments, available-for-sale financial assets, and long-term equity investments.

Journal Pre-proof

In practice, financing asset sales are substantial (Edmans and Mann, 2019). An exhaustive survey conducted by Campello et al. (2010) shows that most financially constrained firms in the United States (US), Europe, and Asia engaged in asset sales to fund their operations during the recent global financial crisis. Similarly, Arnold et al. (2018) note that the average proceeds of fixed asset sales for US manufacturing firms in COMPUSTAT were about 44% of their net average amount of newly-issued equity from 1971 to 2010. Furthermore, Hovakimian and Titman (2006) and Borisova and Brown(2013) find that the proceeds from asset sales significantly affect the investment expenditures of financially constrained firms. In addition, Desai and Gupta (2019) show that the average value of financing asset sales accounts for 4.5% of total assets, while that of the financing by security issuance takes 12.3% of total assets .

Nevertheless, how to convert R&D inputs, including asset sale proceeds, into future growth and income is a central issue regarding R&D smoothing with financing asset sales. A firm's innovation outcomes are determined not only by its R&D inputs but also (and more importantly) by its innovation efficiency (Fu, 2012; Hottenrott and Peters, 2012).³ As pointed out by Zhang(2015), R&D investment may threaten a firm's existence and development if a large proportion of the resources it devotes to innovation projects are unproductive. Thus, R&D-intensive firms are more likely to transfer asset sale proceeds to R&D investments if their innovation efficiencies are high. Innovation efficiency may therefore provide a compelling motivation for R&D smoothing through financing asset sales.

While many studies have investigated innovation activities, studies on R&D smoothing with financing asset sales have moved to the forefront of innovation management research (e.g., Brown and Petersen(2011; 2015); Borisova and Brown(2013)). However, the literature has not yet identified the determinants of financing asset sales for R&D smoothing. Specifically, how financial constraints and innovation efficiency affect the R&D smoothing with financing asset sales among R&D-intensive firms has not been fully explored.

This study attempts to shed light on these issues by providing empirical evidence from R&D-intensive firms listed on China's stock markets. This paper differs from previous studies in that it examines the determinants of converting asset sale proceeds to R&D investment and the impacts of innovative capabilities on investment

³ Similarly, Lach and Schankerman (1989) and Kang et al. (2017)find that R&D activity is determined jointly by economic factors and technological capability.

Journal Pre-proof

adjustment and financing decisions. Specifically, this study contributes to the literature as follows. First, this appears to be the first study, to the best of our knowledge, to consider the interaction effect of financial constraints and innovation efficiency in R&D-asset sale sensitivity comprehensively. Second, we consider both operating assets sale and financial assets sale and compare their impacts on R&D protection, since these two types of asset sales have different financing costs due to asset specificity. Finally, we consider different levels of technological innovation efficiency on R&D-asset sale sensitivity. Our study connects asset divesture theories to R&D investment management from the perspective of R&D smoothing with asset sales, and helps understand the investment adjustment and financing decision behavior of R&D-intensive firms in emerging markets like China.

Our exploration of R&D smoothing with financing asset sales produces four major findings. First, we reveal that R&D-intensive firms have incentives to protect their R&D investments by selling operating and financial assets. Second, we find that financial constraints, in sharp contrast to innovation efficiency, have negative moderating effects on the relationship between asset sales and R&D investments. Third, the results show that financially constrained firms with high innovation efficiency are more inclined to protect their R&D investments with financing asset sales. Finally, the study reveals that innovation efficiency plays a dominant role in R&D smoothing with financing asset sales, while financial constraints may serve as an alternative motivation for asset sales.

The rest of this paper proceeds as follows. Section 2 reviews the relevant literature. Section 3 discusses the study's theoretical framework and presents the hypotheses. Section 4 explains the study's methodology. Section 5 presents the data, indices, and empirical results. Section 6 conducts a robustness test. Finally, Section 7 concludes the paper.

2. Literature review on R&D financing

It is commonly argued that R&D-intensive firms have strong incentives to maintain a smooth path of R&D investment owing to the multiple types of adjustment costs involved (Borisova and Brown, 2013; Brown and Petersen, 2015; Kang et al., 2017). First, R&D investments are accompanied by substantial expenses for firing or rehiring

highly-trained scientists, engineers and other specialists, who often require a large amount of firm-specific training (Lach and Schankerman, 1989; Himmelberg and Petersen, 1994; Hall, 2002; Peter and Taylor, 2017). The valuable firm-specific knowledge embedded in human capital is also lost when employees are released. Moreover, critical proprietary information will likely be disseminated to competitors, undermining innovation value. Second, R&D investment is associated with high sunk costs, as it is largely unrecoverable (Máñez et al., 2009). These costs usually constitute entry and exit barriers for innovation activities (Ganter and Hecker, 2013; Kang et al., 2017). Third, innovation project termination may demoralize team members and raise concerns about job security (Balachandra et al., 1996). Finally, R&D investment suffers from time-compression diseconomies (Dierickx and Cool, 1989; Sears, 2017; 2018). In other words, it is more efficient for R&D investment to keep consistency over a long period rather than double the total investment over a half-time interval.

Although firms try to smooth R&D investment, it is difficult to obtain externally (especially debt) financing for R&D-driven innovation (Hall, 2002). External financiers such as equity investors and banks are reluctant to invest in R&D projects because of their unique characteristics. First, R&D investments generally involve high risk and uncertainty regarding their final output and economic return (Hall, 2002; Hall and Lerner, 2010; Fernandez, 2017). As pointed out by Ghosal and Ye (2015) and Banerjee and Siebert(2017), various kinds of uncertainties, such as technological, demand (Pindyck 1993a, 1993b), and policy uncertainties (Wang et al., 2017), arise throughout different stages of the R&D process. These uncertainties impede external financing for such projects, as they frequently result in development failures (Banerjee and Siebert, 2017).

Second, most R&D investments are intangible assets, which are highly irreversible and offer very limited collateral value (Almeida and Campello, 2007; Drivera and Guedes, 2012). A lesser ability to pledge collateral increases the cost of external financing and reduces debt capacity (Benmelech and Bergman, 2009), while higher-risk firms are generally required to pledge collateral for bank loans (Berger and Udell, 1990). As noted by Hall (2002), banks and other debt-holders prefer to lend when the project involves tangible (physical) assets rather than R&D investments. Nevertheless, servicing debt usually demands a steady cash flow, which in turn diminishes the cash flow available for future investments (Hall, 2002; Hottenrott and Peters, 2012). The cash flow requirement makes external fundraising difficult and unappealing, as most R&D projects are not immediately productive.

Finally, R&D investment projects are likely to suffer from information asymmetry problems. Innovators generally have information about the probability of technical success and the profitability of their current innovation projects that is superior to the information possessed by potential investors (Hall and Lerner, 2010; Drivera and Guedes, 2012). Firms have incentives to keep secrecy regarding the details of R&D projects to avoid disseminating valuable information to competitors (Mohamed and Schwienbacher, 2016), though it raises their cost of capital because of the lemons premium (Hall, 2002; Balakrishnan et al., 2014). ⁴ Meanwhile, information asymmetries are also exacerbated by accounting rules, the absence of organized innovation markets, and the uniqueness of R&D investments (Aboody and Lev, 2000; Guariglia and Liu, 2014).

Given these financial constraints, R&D-intensive firms may use their internal cash flow or cash holdings for R&D smoothing (Brown and Petersen, 2011; He and Wintoki, 2016; Ahrends et al., 2018; Shao and Xiao, 2019),⁵ as external financing is either unavailable or more expensive than internal financing (Bernini and Montagnoli, 2017). However, internal cash holdings are exhaustible, and cash flow is naturally restricted (Hottenrott and Peters, 2012). In this scenario, firms may be forced to sell non-core assets and utilize the proceeds to maintain their R&D investments (Borisova and Brown, 2013; Brown and Petersen, 2015). Asset sales provide an alternative way to raise capital in order to finance investment opportunities or meet liquidity needs (Edmans and Mann, 2019) when external financing is scarce or costly (Borisova et al., 2013).

According to Borisova and Brown(2013), the average asset sale increases R&D intensity by 6.9% of the sample mean, which is only slightly less than the amount that would be generated (8.8%) by stock issues. Meanwhile, Hovakimian and Titman (2006) find that the sensitivity of investments to asset sale proceeds is significantly stronger for financially constrained firms. Furthermore, Brown and Petersen (2015)

⁴ Admittedly, the relationship between a firm's capital cost and disclosure quality may depend on growth rate thresholds (Dutta and Nezlobin, 2017).

⁵ However, the financing sources for R&D investment remain ambiguous. Some researchers, such as Brown et al.(2012), find that external equity financing plays a major role for R&D investments. Contrariwise, Chay et al. (2015)find that firms use internal funds for R&D investments. Similarly, Sasidharan et al. (2015)fail to find any significant evidence that firms use external equity for R&D financing or engage in R&D smoothing using cash reserves.

find that financially constrained firms favorably allocate cash holdings to buffer R&D investments, even taking the extreme step of allowing their fixed investments to fall.

Although previous studies provide valuable insight into R&D smoothing, most studies seek to identify financial constraints by controlling for R&D smoothing (e.g., Hovakimian and Titman(2006); Brown et al.(2012); Hottenrott and Peters(2012); He and Wintoki(2016)) rather than the issue *per se*. Only a few recent studies, such as Brown and Petersen(2011), Borisova and Brown(2013), and Brown and Petersen(2015), have discussed R&D smoothing with cash holdings or asset sale proceeds. More recently, Kang et al.(2017) have examined the impact of technological capability on the persistence and volatility of R&D investment. However, these studies have neither addressed the determinants of R&D smoothing with financing asset sales nor examined their impacts on the relationship between asset sales and R&D investment protection. This study attempts to fill these gaps by examining the heterogeneous effects of financial constraints and innovation efficiency on R&D smoothing with financing asset sales.

3. Theory and hypotheses

The literature review shows that R&D smoothing is of significant importance for R&D-intensive firms owing to various adjustment costs, including firing or rehiring costs for highly skilled employees, sunk costs, and time-compression diseconomies (Borisova and Brown, 2013; Brown and Petersen, 2015; Kang et al., 2017). However, it is difficult for these firms to smooth their R&D investments, since R&D activities are easily subject to financial constraints due to high uncertainty, asymmetric information, and limited collateral value (David et al., 2000; Hall, 2002; Brown et al., 2012). Facing financial constraints, these firms may choose to protect some value-enhancing investments according to capital adjustment costs while cutting their investment expenditures. For example, Borisova and Brown (2013) show that firms with limited external financing use the funds obtained from tangible assets sale for R&D activities. Similarly, Brown and Petersen (2015) also point out that fixed asset investment is reduced more than R&D investment when firms face severe financial shocks in the financial crisis. Therefore, firms are inclined to allocate more cash flow into intangible assets (e.g. R&D investment) rather than into tangible assets when selling assets.

Journal Pre-proof

As far as asset sales are concerned, financial assets sale rather than operating assets sale plays an important role of financing in corporate financial decisions. Some studies begin to pay attention to the relationship between firm financial asset allocation and R&D innovation investment (Hahn, 2019). Compared to operating assets associated with high asset specificity, it is less costly for firms to raise funds by selling financial assets, since the latter generally trade at a lower liquidity cost in a more active trading market. Financial assets with high liquidity can be used as precautionary savings for R&D investment and alleviate the negative effects of financial constraints on R&D activities. In this regard, financial assets sale is beneficial to R&D investment. Apart from financing facility, however, the excess returns of arbitrage also provide firm a speculative motivation to allocate the proceeds from financial assets sale (Orhangazi, 2008; Demir, 2009). As pointed out by Orhangazi (2008) and Demir (2009), the allocation of financial assets has a "crowding out effect" on other types of investment. In other words, financial assets sale may have adverse impact on R&D investment.

As the particular impact of financial assets sale on R&D protection remains unclear, R&D-intensive firms in China provide a unique experiment for studying this issue. In recent years, many China's non-financial firms have been distracted from their main business and involved in financial and real estate fields. These firms allocate excessive investment in financial assets to obtain higher returns than their main business, so that the investment income of financial assets have gradually become an important part of profits (Zhang and Zhang 2016). Meanwhile, the proportion of corporate financial asset investment has also increased greatly, especially after 2012 when the China Securities Regulatory Commission issued the *Guidelines for the Supervision and Administration on Listed Companies No.2-Supervision and Administration Requirements for Listed Companies on the Management and Use of Raised Funds.* The guidelines allow listed firms to use idle raised funds to purchase investment products with high security and good liquidity. As a result, the average holdings of financial assets by non-financial listed firms have increased greatly in China, which is presented in Figure A.1 in Appendix.

Overall, operating assets sale and financial assets sale may have heterogeneous effects on R&D protection, since these two types of asset sales have different liquidity costs. For the comparison purpose, this study considers both operating assets and financial assets sales and proposes the general hypothesis as follows. **H1.** Research and development-intensive firms have incentives to protect their value-enhancing R&D investments by selling assets with relatively low adjustment costs, such as operating assets and financial assets.

However, R&D-intensive firms may face a R&D protection dilemma when they are financially constrained. On the one hand, financially constrained firms may tend to protect their R&D investments by selling non-core assets, as this provides alternative funds when external financing is especially scarce or costly (Borisova and Brown, 2013). On the other hand, financially constrained firms may be forced to reduce their R&D investments, as they typically use the proceeds from asset sales to meet debt obligations (Shleifer and Vishny, 1992) rather than retain the sale proceeds within the firm (Lang et al., 1995). Therefore, the competing hypotheses below are proposed:

H2a. Financially constrained firms are more inclined to protect their R&D investments with financing asset sales.

H2b. *Financially constrained firms are forced to reduce their R&D investments when they raise capital through financing asset sales.*

Nevertheless, how to efficiently convert the proceeds from asset sales into innovation outputs is a fundamental issue for firms engaged in R&D smoothing through financing asset sales. The conversion process depends heavily on innovation efficiency, which directly determines the innovation outputs achieved from a given set of innovation inputs (Fu, 2012; Guan and Chen, 2012), as dedicating more innovation inputs cannot, alone, guarantee innovation outcomes(Zhang 2015). As pointed out by Cruz-Cázares et al.(2013), the key to increasing firm performance is technological innovation efficiency. Gao and Chou (2015) find that innovation efficiency can improve the value of multinational firms. Similarly, Hirshleifer et al. (2013)find that innovative efficiency are more likely to achieve innovation outputs, and thus have a strong incentive to protect their ongoing R&D projects. This leads to the final hypothesis:

H3. Research and development-intensive firms with higher innovation efficiency are more likely to sell assets to protect their R&D investments.

4. Methodology

4.1 Baseline model of R&D smoothing

Inspired by the seminal work of Brown et al. (2009)and Brown and Petersen(2011), this study constructs dynamic R&D models with financial variables to examine the R&D smoothing with financing asset sales. These models are based on an Euler equation proposed by Bond and Meghir (1994) under the assumption of quadratic adjustment costs for productive assets. The Euler equation is a structural model created through derivation from the dynamic optimization "Euler condition" for imperfect competition. It captures the impact of current expectations of future profitability on current decisions (Bond et al., 2003; Brown et al., 2009; Whited and Wu, 2006).

Incorporating asset sale proceeds (ASale) for R&D smoothing, the Euler equation leads to the following empirical specification:

$$Rd_{i,t} = \alpha_1 ASale_{i,t} + \alpha_2 IE_{i,t-1} + \alpha_3 FC_{i,t-1} + \alpha_4 Rd_{i,t-1} + \alpha_5 Rd_{i,t-1}^2 + \alpha_6 Growth_{i,t} + \alpha_7 Cf_{i,t} + \alpha_8 Debt_{i,t} + \alpha_9 Stk_{i,t} + \alpha_{10} Size_{i,t} + \mu_i + \nu_t + \varepsilon_{i,t}$$

$$ASale = OP \text{ or } In\nu$$
(1)

where $Rd_{i,t-1}$ reflects the R&D spending of firm *i* at period t - 1, while $Rd_{i,t-1}^2$ denotes the quadratic adjustment costs of R&D investment. *ASale* represents the proceed from each type of assets sales, namely, operating assets sale (*OP*) or financial assets sale (*Inv*). Innovation efficiency (*IE*) and financial constraint (*FC*) are two main explanatory variables. Following Borisova and Brown (2013), this study also considers financial variables at the firm level, such as cash flow (*Cf*), net debt issues (*Debt*), and net stock issues (*Stk*). Meanwhile, investment opportunities (*Growth*) and firm size (*Size*) are also included to control for investment demand. Eq. (1) also incorporates firm-fixed effect (μ_i) and time-specific effect (v_t) to control for unobserved, time-invariant factors, and aggregate changes respectively. Finally, $\varepsilon_{i,t}$ is the idiosyncratic error term. The constant terms in the regression models are omitted for the sake of brevity. All financial variables are scaled by total assets at the beginning of the period. A detailed explanation of these variables is provided in Table A.1 in Appendix.

The parameters in Equation (1) can be interpreted as functions of those in the original

optimization problem of the Euler equations (Brown et al., 2009; Guariglia and Liu, 2014). Under the assumption of quadratic adjustment costs in the Euler condition, the expected coefficient on lagged R&D should be positive and that on the quadratic term should be negative. As aforementioned, the coefficients on *OP* and *Inv* should be positive for firms that choose to protect their R&D investments by selling assets.

Meanwhile, the coefficients of *Debt* and *FC* are predicted to be negative.⁶ Firms with high debt ratios are more likely to reduce their R&D expenditures because R&D investments may evaporate in times of financial distress (Bhagat and Welch, 1995). Moreover, cash flow (*Cf*) and stock issues (*Stk*) should share a positive relation with R&D, while a similar pattern is expected to emerge in the relationships between R&D and investment opportunities (*Growth*), firm size (*Size*), and innovation efficiency (*IE*).

4.2 Moderating effect of financial constraints

Since the seminal work of Fazzari et al.(1988), a large body of literature has focused on the identification of financial constraints (e.g., Brown et al.(2012); Chen and Chen(2012); Foley-Fisher el al.(2016); Moshirian et al.(2017)). However, there is no consensus on how to measure financial constraints properly, and a diverse range of approaches has been applied (Erel et al., 2015; Chen et al., 2017). The main measures include the Kaplan–Zingales (KZ) index (Kaplan and Zingales, 1997), the Whited– Wu (WW) index (Whited and Wu, 2006), and the investment-cash flow sensitivity (ICFS) method (Fazzari et al., 1988).⁷ As pointed out by Erel et al.(2015), each measure has limitations while also providing valuable insight into how to assess financial constraints.

As an alternative, the Euler equation approach is proposed by Howell(2016). This approach, based on the Euler equation, has several advantages over the other measures(Howell 2016). First, this approach addresses the critique of Kaplan and Zingales (1997) that ICFS is not a reliable measure of financial constraints, as their claim has not been substantiated in the dynamic multiperiod setting of the Euler equation. Second, this approach requires no further information regarding firms'

⁶ The literature does not provide a clear determination of the relationship between R&D investment and debt. Some studies, such as Bhagat and Welch (1995) and Wang(2017), find that R&D investments are negatively associated with indebtedness. By contrast, Guney et al. (2017)argue that firms may use credit lines to finance their R&D investments. Mann (2018) finds that patenting companies conduct significant debt financing for innovation.

⁷ The readers are referred to Khatami et al. (2015)and Mulier et al. (2016)for excellent literature reviews on financial constraints.

market value or dividends, and the assumptions required to estimate the Euler equation are less restrictive. Finally, this approach controls for all expectations of future influences on current investment decisions.

Incorporating a set of firm characteristics to represent financial constraints, the Euler equation is specified as follows:

$$I_{i,t} = \mu_1 I_{i,t-1} + \mu_2 Sales_{i,t-1} + \Omega * Cf_{i,t-1} + \eta_t + \xi_{i,t}$$
(2)

with

$$\Omega = v_1 Size_{i,t} + v_2 Age_{i,t} + v_3 Lev_{i,t} + v_4 cash_{i,t}$$
(3)

where $I_{i,t}$ is the investment expenditure of firm *i* in period *t*, $Sales_{i,t-1}$ the net revenue generated from the selling of products, goods and services in the previous year, $Cf_{i,t-1}$ the net cash flow received from operating activities in the previous period, $Size_{i,t}$ the firm size measured by the firm's total assets, $Age_{i,t}$ firm age, $Lev_{i,t}$ the ratio of long-term liabilities to total assets, and $cash_{i,t}$ the cash holding.

Firm size, age, cash flow, and leverage are generally believed to capture a firm's financial constraints well (e.g., Rauh(2006); Hadlock and Pierce(2010); Mulier et al. (2016) and Howell(2016)). Firm size and age are the most important factors in any measure of financial constraints (Beck et al., 2006; Fee et al., 2009; Hadlock and Pierce, 2010). A firm's age is closely associated with its informational opacity, which affects its creditworthiness and capital costs due to information asymmetry and adverse selection (Hyytinen and Pajarinen, 2008). Firm size also matters for credit extension, which involves overcoming problems of information asymmetry and insufficient collateral (Bernanke et al., 1996). Smaller firms tend to face more severe financial constraints because they are more likely to have volatile growth patterns and earnings (Howell 2016) and less of a proven track record(Schiantarelli 1996). Meanwhile, cash flow is crucial for repaying debt and therefore determines the firm's debt capacity (Mulier et al., 2016). Finally, less-levered firms are more likely to be financially unconstrained because they are stable and profitable (Strebulaev and Yang, 2013), and thus have lower solvency risk (Mulier et al., 2016).⁸ Unlike Howell(2016), this study does not include other variables, such as export and subsidy variables,

⁸ Admittedly, there is a controversy regarding the relationship between financial constraints and indebtedness. As pointed out by Bessler et al.(2013), for example, most zero-leverage (i.e., extreme debt-conservative) firms are financially constrained because they have insufficient debt capacity, while only a small number of financially unconstrained firms with high profitability and dividend payment may deliberately choose a zero-leverage policy by avoiding debt financing. Readers are referred to Bessler et al. (2013)and Dang(2013) for detailed literature reviews.

owing to data unavailability issues.

The estimated coefficients for the v's in Equation (3) are used to calculate the firm-specific financial constraint score, which is based on the firm's characteristics. Although the estimated coefficients remain constant over the full sample period, the degree of a firm's financial constraints is time-varying because its features change over time. The firm-level financial constraint score Fi, t is obtained as follows:

$$F_{i,t} = v_1 Size_{i,t} + v_2 Age_{i,t} + v_3 Lev_{i,t} + v_4 cash_{i,t}$$
(4)

A firm is considered to be financially constrained if $F_{i,t} > 0$, while a higher score indicates more severe financial constraints. For ease of interpretation, a dummy variable *FC* is set equal to one if $F_{i,t} > 0$ and zero otherwise. In other words, firms are sorted into financially constrained and unconstrained groups according to their financial constraint scores.

The dummy variable FC is introduced to examine the moderating effect of financial constraints on the relationship between asset sales and R&D protection. Incorporating the interaction term of financial constraint and proceeds from asset sales, Equation (1) is reformulated as follows:

$$Rd_{i,t} = \gamma_1 ASale_{i,t} + \gamma_2 Asale_{i,t} \times FC_{i,t-1} + \gamma_3 FC_{i,t-1} + \gamma_4 IE_{i,t-1} + \gamma_5 Rd_{i,t-1} + \gamma_6 Rd_{i,t-1}^2 + \gamma_7 Growth_{i,t} + \gamma_8 Cf_{i,t} + \gamma_9 Debt_{i,t} + \gamma_{10} Stk_{i,t} + \gamma_{11} Size_{i,t} + \mu_i + \nu_t + \varepsilon_{i,t}$$

$$ASale = OP, or Inv$$
(5)

where coefficient γ_2 represents the magnitude of the moderating effect of financial constraints on financing asset sales (operating assets or financial assets). A positive coefficient γ_2 represents that financially constrained firms protect their R&D investments with the proceeds from asset sales; otherwise, the firms reduce their R&D investments when raising capital through asset sales.

4.3 Moderating effect of innovation efficiency

A firm's innovation efficiency is generally understood as its capacity to generate innovation outputs per unit of innovation inputs (Li, 2009; Hirshleifer et al., 2013; Gao and Chou, 2015; Knut et al., 2017). Numerous studies have adopted a wide range of indicators for innovation inputs, such as innovation expenditure (Fu, 2012; Knut et al., 2017), R&D capital (Hirshleifer et al., 2013; Griffin et al., 2018), and R&D manpower (Wang and Huang, 2007; Fu, 2012). Some studies have also proposed a

variety of innovation output indicators, including patents and patent citations (Seru, 2014; Gao and Chou, 2015), innovative sales(Fu 2012), and product innovation (Cruz-Cázares et al., 2013). Unfortunately, there is no widely accepted measurement, as the mixed use of different types of indicators may lead to ambiguous results (Cruz-Cázares et al., 2013).

Following Hirshleifer et al. (2013)and Griffin et al.(2018), this study uses the share of a firm's number of patent filings scaled by its R&D expenditures as the measure of innovation efficiency. This measure indicates the firm's capability to transform innovation input (i.e., R&D expenditures) into innovative output (i.e., patents).⁹ Although patent citations provide valuable information regarding the quality of innovative output (Gao and Chou, 2015), this study does not include these data, as they are unavailable in China.

Instead, this study considers three types of patents in China, namely, invention, utility model, and design. These three types of patents reflect the technological and economic significance of patents as they help distinguish between breakthrough innovations and less-innovative or superficial novelties.¹⁰ To provide the broadest coverage, this study constructs three indicators of innovation efficiency using the shares of the firm's patent application filings for inventions, utility models, and designs divided by its R&D expenditures respectively. These indicators of innovation efficiency, denoted by *IE'*, *IE''*, and *IE''* correspondingly, capture innovation qualities well. It is worth noting that this study mainly focuses on inventions and utility models, as they both involve "new technical solutions", which is not involved in designs.¹¹

The innovation efficiency indicator IE(IE', IE'' and IE'') is included to examine the moderating effect of innovation efficiency on the relationship between asset sales and R&D protection. Incorporating the interaction term of innovation efficiency and proceeds from asset sales, Equation (1) is reformulated as follows:

$$Rd_{i,t} = \delta_1 ASale_{i,t} + \delta_2 Asale_{i,t} \times IE_{i,t-1} + \delta_3 IE_{i,t-1} + \delta_4 FC_{i,t-1} + \delta_5 Rd_{i,t-1} + \delta_6 Rd_{i,t-1}^2 + \delta_6 Rd_{i,t-1}$$

⁹ Although patents, like other indicators, are subject to criticism, they are one of the most readily available and reliable measures of innovation output (Li, 2009; Gao and Chou, 2015).

¹⁰ According to China's patent system, patent quality varies across three categories (Li, 2009). Invention patents represent the most valuable and technologically sophisticated innovation outputs. Utility models are less significant than inventions in terms of technological innovation but are more innovative than designs. In other words, designs contain less technological innovation and indicate superficial novelty. China's "invention patents" are equivalent to US "utility patents", while "design patents" in China are similar to their US counterparts. However, China's "utility models" have no US equivalent (Christodoulou et al., 2018). Readers are referred to *the Patent Law of the People's Republic of China* for definitions of inventions, utility models, and designs.

¹¹ Invention patents in China are relatively sparse compared to utility model and designs patents.

$$\delta_7 Growth_{i,t} + \delta_8 C f_{i,t} + \delta_9 Debt_{i,t} + \delta_{10} Stk_{i,t} + \delta_{11} Size_{i,t} + \mu_i + \nu_t + \varepsilon_{i,t}$$
(6)

$$ASale = OP, or Inv, and IE = IE', IE'' or IE'''$$

where coefficient δ_2 represents the magnitude of the moderating effect of innovation efficiency on asset sales, and $\varepsilon_{i,t}$ is the residual. The coefficient δ_2 is expected to be positive if firms with high innovation efficiency are more likely to protect their R&D investments by financing asset sales.

This study examines the combined impact of financial constraints and innovation efficiency by incorporating the interaction term of financial constraint and proceeds from asset sales and a triple interaction term of innovation efficiency, financial constraints, and asset sale proceeds into Equation (6) and reformulates it as follows: $Rd_{i,t} = \tau_1 ASale_{i,t} + \tau_2 Asale_{i,t} \times FC_{i,t-1} \times IE_{i,t-1} + \tau_3 Asale_{i,t} \times FC_{i,t-1} + \tau_4 Asale_{i,t} \times IE_{i,t-1} + \tau_5 FC_{i,t-1} + \tau_6 IE_{i,t-1} + \tau_7 Rd_{i,t-1} + \tau_8 Rd_{i,t-1}^2 + \tau_9 Growth_{i,t} + \tau_{10} Cf_{i,t} +$

$$\tau_{11} Debt_{i,t} + \tau_{12} Stk_{i,t} + \tau_{13} Size_{i,t} + \mu_i + \nu_t + \varepsilon_{i,t}$$
(7)

$$ASale = OP$$
, or Inv , and $IE = IE'$, IE'' or IE'''

where coefficient τ_2 represents the extent to which financial constraints and innovation efficiency jointly impact the moderation between asset sales and R&D investment. The coefficient τ_2 is predicted to be positive if financially constrained firms with high innovation efficiency sell assets to protect their R&D investments.

5. Empirical analysis

5.1 Indices and data description

This study selects Chinese A-share listed companies with significant R&D investments from 2009 to 2016 to examine their R&D smoothing with financing asset sales.¹² These companies are listed on Chinese exchanges and primarily available for domestic investors. Most of these firms are in technology-intensive industrial sectors, including the chemical material industry, pharmaceutical industry, general equipment industry, specialized equipment industry, automotive industry, computer industry, and software and information technology services industry.¹³ In order to select firms with

¹² The Chinese government implemented the *New Accounting Standards* in 2007 to regulate information disclosure with respect to investment accounts. This study does not include the data before 2009 to avoid the distortions caused by accounting standards changes and the global financial crisis in 2008.

¹³ According to the *Statistics Catalogue of High-technology Industry Classifications* issued by the National Bureau of Statistics of China in 2002, China's technology-intensive industrial sectors are concentrated in electronics and communications equipment manufacturing, electronic computer and office equipment

substantial R&D expenditure, this study excludes any firm without at least four positive R&D observations in the sample period (i.e., 50% during the period).¹⁴ In so doing, there are at least five listed companies within each sector. In addition, this paper also eliminates the ST and *ST firms. A total of 2142 observations remain to be analyzed after preliminary screening. These datasets meet the sample requirement as all firm-level datasets should be standardized by the mean and standard deviation of their industries.

Firms' financial data are retrieved from the Wind Financial Database, which the literature has often used (e.g., Zheng et al.(2018); Jiang and Yuan(2018)), while patent information is collected from the official website of the State Intellectual Property Office of China (SIPOC). Operating assets sale is calculated according to the cash received from the sale of fixed assets, intangible assets and other long-term assets, while financial assets sale is the cash received from the investment recovery in the cash flow statement of the firm.¹⁵ All continuous variables are winsorized at the 1st and 99th percentiles to mitigate the impact of possible outliers. The detailed information on the variables is presented in Table A.1 in Appendix, and summary descriptive statistics for the firms' financial data are displayed in Table 1.

	Full	sample	Financially un	nconstrained	Financially	y constrained	Mean difference		
	(N - 2142)		$(N \equiv 3)$	(88)	$(N \equiv$:1/54)	between two groups		
Variable	Mean	Standard Deviation	Mean	Standard Deviation	Mean	Standard Deviation	t-statistic	Z-statistic	
$Rd_{i,t}$	0.0279	0.0238	0.0223	0.0197	0.0291	0.0249	-0.4884	0.033	
$OP_{i,t}$	0.0028	0.0107	0.0004	0.0010	0.0033	0.0141	-1.5081*	-3.196**	
Inv _{i,t}	0.0554	0.3594	0.0415	0.1392	0.0801	0.3511	-0.7993*	-2.487*	
$Cf_{i,t}$	0.0393	0.0605	0.0412	0.0687	0.0407	0.0602	-1.0292	-1.5013	
Growth _{i,t}	0.3094	3.2748	0.2569	0.1472	0.4004	3.9077	-0.1797	4.003***	
Stk _{i,t}	0.0968	0.2806	0.0436	0.1739	0.1234	0.2682	-2.0850**	-2.632**	
Debt _{i,t}	0.0429	0.0983	0.0412	0.0927	0.0509	0.1012	-0.0994*	-1.010*	
Size _{i,t}	21.8869	1.1704	22.0122	1.2003	20.9353	0.9987	-1.0273*	-2.501*	
$Lev_{i,t}$	0.3072	0.2001	0.2002	0.1731	0.3790	0.1934	-8.0211***	-8.9750***	
$IE_{i,t-1}$	0.4022	0.9980	0.3410	0.4879	0.3994	0.8011	-0.1610	0.0328	

 Table 1. Descriptive statistics for financial variables

manufacturing, pharmaceutical manufacturing, and medical equipment manufacturing.

¹⁴ The results in general are robust and consistent under more restrictive conditions.

¹⁵ According to CASs, cash received from investment recovery refers to the cash received by the company from the sale, transfer or maturity of transactional assets (other than cash equivalents), held-to-maturity investments, available-for-sale financial assets, long-term equity investments, and investment real estate.

Notes: In this table, *N* stands for the number of observations. The mean differences between financially constrained and unconstrained groups are obtained using the t-test and the Wilcoxon (Mann–Whitney) rank-sum test respectively. ***, **, and * denote significance at the 1%, 5%, and 10% levels respectively. Definitions of variables are provided in Table A.1 in Appendix.

As shown in Table 1, the average R&D intensity (*Rd*) is 2.79%, while the mean of the operating assets sales ratio (*OP*) is 0.28% and that of financial assets sales (*Inv*) is 5.54%. Financial assets sales are more salient in Chinese R&D-intensive firms than operating assets sales and are nearly double R&D intensity. In other words, not all proceeds from asset sales are put into R&D investment. Firms with financial assets sales sales may have certain motivation for speculative profits. For example, the average holdings of financial assets by Chinese non-financial listed firms have increased significantly according to the secular trend of financial asset allocation presented in Figure A.1 in Appendix. The share of cash flow from operating activities (*Cf*) is 3.93%, which is almost equivalent to that of debt issues (4.29%) and is half that of stock issues (9.68%). Nevertheless, asset sales play an important role in raising funds, accounting for nearly 32.5% of total fundraising

Table 1 also shows that most R&D-intensive firms in China are financially constrained, even though they are listed firms on Chinese stock markets. This result is consistent with the results of previous studies regarding the characteristics of R&D investments. The mean differences between financially constrained and unconstrained groups show that these two groups are statistically significant with respect to net stock issues (*Stk*), and leverage (*Lev*). Financially constrained firms generally have greater leverage, and stock issues. However, there is no significant difference between the two groups in terms of other firm features, such as cash flow (*Cf*), innovation efficiency (*IE*), and R&D investment (*Rd*). These findings somewhat alleviate the endogeneity concern regarding firm performance, innovation efficiency, and R&D investment because, intuitively, they affect each other. Moreover, the two groups are barely significant in terms of firm size (*Size*) and net debt issues (*Debt*).

5.2 Generalized method of moments estimation results

Dynamic panel data models may suffer from endogeneity, though some variables are uncorrelated. This study employs the system generalized method of moments (GMM) estimator developed by Arellano and Bover (1995) and Richard and Bond (1998) to address the potential endogeneity of the regression variables.¹⁶ The system-GMM

¹⁶ The GMM model can also address temporal specific and individual effects (Bandyopadhyay and Barua, 2016)

Journal Pre-proof

estimator regards all variables, including differenced and lagged variables, as potentially valid instruments if they are uncorrelated with the error term (Guney et al., 2017). The GMM model controls for endogeneity via data transformation (Ullah et al., 2018), which includes first-difference transformation (one-step GMM) and second-order transformation (two-step GMM). To avoid potential data loss associated with the one-step GMM (Ullah et al., 2018), this study uses the two-step GMM method to estimates the regressions with lagged dependent variables up to period t - 4 as instruments.¹⁷ The downward-biased standard errors of the two-step GMM for small samples are also calibrated via Windmeijer's (2005) finite-sample correction. The two-step GMM estimators are reported in Tables 2 and 3 respectively.

This study also uses the Sargan test of over-identifying restrictions to assess instrument validity and the Arellano–Bond test to examine autocorrelation in residuals (Ullah et al., 2018). According to the Sargan test, the instruments included in the econometric models are exogenous if the test turns out to be statistically insignificant. For the Arellano–Bond test, the GMM estimator is valid if a first-order serial correlation (i.e., m1) is observed and there is no second-order serial correlation (i.e., m2) in the residuals. The results of the Sargan and Arellano–Bond tests are reported in Tables 2 and 3 respectively. The results show that the GMM estimators in this study pass the diagnostic tests and are therefore valid.

5.3 R&D protection via asset sales

This section uses the two-step GMM to estimate the regressions and analyzes the impacts of financial constraints and innovation efficiency on R&D protection by asset sales. Tables 2 and 3 report the GMM estimators with respect to R&D protection with proceeds from operating assets sale and financial assets sale respectively.

As shown in Tables 2 and 3, the coefficients on lagged R&D are significantly positive at the 1% level, while those on their quadratic terms are significantly negative. This finding is consistent with the expectation that there are quadratic adjustment costs in R&D investment. Meanwhile, the coefficients associated with innovation efficiency (*IE*), investment opportunities (*Growth*), and cash flow (*Cf*) are significantly positive, while that of net debt issues (*Debt*) and financially constraint (*FC*) are significantly

as well as the dynamic panel bias arising from the correlation between firm-specific fixed effects and lagged dependent variables (Guney et al., 2017).

¹⁷ This study also tried the one-step GMM and found that the model could hardly pass the over-identification test.

negative. These findings imply that R&D investment is determined not only by firms' financial conditions but also by their previous R&D investments, innovation efficiency, and market conditions.

More interestingly, R&D protections with operating assets sales and financial assets sales are generally salient after controlling for other financial variables. The coefficients associated with asset sale proceeds from operating assets (*OP*) and from financial assets (*Inv*) in Eq. (1) are significantly positive at the 1% level, except the coefficient of *Inv* for design case, which is only significant at the 10% level. These results support Hypothesis 1 that firms are generally inclined to protect their R&D investments by selling their operating assets and financial assets. The results also support previous studies suggesting that R&D-intensive firms tend to use available internal financing, rather than external financing, for their R&D investments.

Tables 2 and 3 also show that the coefficients on the interaction terms of financial constraints and operating assets sales are significantly negative for both invention and utility model in Eq. (5) (at level of 1% and 10% respectively). However, their counterparts are negative but insignificant at conventional levels in the design case. These findings reject Hypothesis 2a and therefore support Hypothesis 2b, implying that financial constraints have a negative moderating effect on R&D protection with financing asset sales, especially for firms with cutting-edge innovation. These firms may face more uncertainty and higher risk of R&D failure, and therefore, the negative effects of financial constraints are more prominent (Kamien and Schwartz, 1978). This finding is consistent with Campello et al. (2010) and Zhang (2015) that financially constrained firms are more likely to reduce, rather than protect, their R&D investments, even if they choose to raise capital by selling assets. This finding also supports the view of Borisova and Brown (2013) that the R&D financing consideration is not the primary motivation for asset divestitures.

Tables 2 and 3 further show that most of the coefficients on the interaction terms of IE and OP (or Inv) are statistically significant for the three innovation types. The positive coefficients in Table 2 suggest that innovation efficiency has significantly positive moderating effects on R&D protection by operating assets sales, supporting Hypothesis 3. By contrast, the coefficients on the interaction terms of innovation efficiency and financial assets sale in Table 3 are significantly positive for the cases of innovative patents (invention and utility model), while the one in the case of

superficial novelty (design) is insignificant. This finding suggests that only firms with high innovation efficiency in core patents are inclined to protect their R&D investments with financial assets sale, partially supporting Hypothesis 3. This finding implies that firms with high efficiency in superficial innovation have less incentive to protect their R&D investments by selling financial assets, even though the financing cost of financial assets sale is lower than that of operating assets sale. The comparison results between two types of asset sales show that R&D smoothing with asset sales is partially influenced by innovation types.

The coefficients on the triple interaction terms in Tables 2 and 3 reflect the joint impact of financial constraints and innovation efficiency on R&D protection with asset sale proceeds. Tables 2 shows that the coefficients are significantly positive at the 1% level in the three innovation cases. Similarly, the coefficients on the triple interaction terms in Table 3 are significantly positive at the 1% level for invention and utility cases, while the coefficient for design case is insignificant. These findings suggest that financially constrained firms with high innovation efficiency are generally inclined to increase their R&D investment with proceeds from asset sales. In other words, innovation efficiencies in these scenarios (except the design case with financial assets sale) are large enough to offset the adverse impacts of financial constraints on R&D protection, so that financially constrained firms with adverse impacts of financial constraints on R&D protect their R&D investments with asset sales.

In general, both operating assets sales and financial assets sales have positive impacts on R&D investment. However, R&D protection with operating assets sale is more salient than the one with financial assets sale. As shown by Tables 2 and 3, for example, the coefficients on operating assets sale and its interaction terms are more significant and larger than those coefficients on financial assets sale. This finding suggest that operating assets sale is mainly to raise funds for R&D protection while financial assets sale has some arbitrage motivations, other than financing facility, to be satisfied (Orhangazi, 2008; Demir, 2009). R&D financing consideration is probably a primary motivation for operating assets sale rather than financial assets sale. Given that operating assets sale has a relatively higher liquidity cost due to asset specificity, these findings also indicate that only truly innovative firms would like to adopt the R&D smoothing strategy with operating assets sales at higher financing costs. Journal Pre-proof

	Invention $(IE = IE')$					Utility mode	el (IE = IE''))	Design $(IE = IE^{''})$				
Variable	Eq. (1)	Eq. (5)	Eq. (6)	Eq. (7)	Eq. (1)	Eq. (5)	Eq. (6)	Eq. (7)	Eq. (1)	Eq. (5)	Eq. (6)	Eq. (7)	
$Rd_{i,t-1}$	0.576 ^{****} (92.18)	0.488 ^{***} (101.10)	0.492 ^{***} (93.74)	0.531 ^{***} (107.22)	0.511 ^{***} (112.70)	0.492 ^{***} (86.17)	0.479 ^{***} (94.12)	0.519 ^{***} (80.12)	0.417 ^{***} (121.16)	0.423 ^{***} (106.23)	0.486 ^{***} (103.12)	0.505 ^{***} (72.13)	
$Rd_{i,t-1}^2$	-0.047 ^{***} (-20.19)	-0.038 ^{***} (-13.14)	-0.033 ^{***} (-20.44)	-0.049 ^{***} (-17.52)	-0.042 ^{***} (-21.13)	-0.050 ^{***} (-16.00)	-0.044 ^{****} (-19.27)	-0.039 ^{***} (-17.33)	-0.043 ^{***} (-15.09)	-0.040 ^{***} (-16.62)	-0.035 ^{***} (-19.92)	-0.042 ^{***} (-15.78)	
$OP_{i,t}$	0.033 ^{***} (12.26)	0.018 ^{**} (2.39)	0.315 ^{***} (7.76)	0.043 (0.88)	0.047 ^{***} (8.65)	0.055 [*] (1.26)	0.083 ^{**} (4.45)	0.244 (0.69)	0.041 ^{***} (9.17)	0.318 ^{**} (2.79)	0.072 (0.33)	0.015 [*] (1.29)	
$OP_{i,t} \times FC_{i,t-1}$		-0.310 ^{***} (-5.32)	0.110***	-0.008** (-3.54)		-0.030* (-1.22)		-0.015 ^{***} (-6.00)		-0.045 (-1.31)	0.00 <***	-0.316 (-0.33)	
$OP_{i,t} \times IE_{i,t-1}$			0.118 (7.74)				0.553 (14.22)				0.006 (8.10)		
$OP_{i,t} \times FC_{i,t-1} \times IE_{i,t-1}$				0.313*** (6.31)				0.090^{***} (8.41)				0.111 ^{**} (3.35)	
$FC_{i,t-1}$	-0.116 ^{***} (-12.35)	-0.052*** (-17.01)	-0.108 ^{***} (-13.34)	-0.088 ^{***} (-15.16)	-0.100 ^{***} (-17.31)	-0.089 ^{***} (-12.90)	-0.104 ^{****} (-15.77)	-0.077 ^{***} (-14.46)	-0.075 ^{***} (-16.29)	-0.080 ^{***} (-12.72)	-0.058*** (-10.35)	-0.041 ^{****} (-13.05)	
$IE_{i,t-1}$	0.401 (17.12)	0.348 (20.13)	0.329 (20.53)	0.288 (19.80)	0.316 (18.42)	0.272 (14.25)	0.209 (15.11)	0.278 (18.40)	0.323 (19.31)	0.300 (15.37)	0.255 (11.15)	0.302 (12.57)	
$Growth_{i,t}$	0.050^{***} (11.21)	0.041 ^{***} (10.25)	0.319 ^{***} (17.31)	0.030 ^{***} (13.11)	0.033 ^{***} (8.31)	0.033 ^{***} (9.31)	0.035^{***} (10.31)	0.031 ^{***} (8.65)	0.039 ^{***} (10.00)	0.310^{***} (6.11)	0.018 ^{***} (7.06)	0.315 ^{***} (8.97)	
$Cf_{i,t}$	0.200^{***} (25.41)	0.209 ^{***} (22.72)	0.235 ^{***} (21.84)	0.117 ^{***} (19.70)	0.097 ^{***} (17.32)	0.131 ^{***} (19.45)	0.084 ^{***} (18.32)	0.107^{***} (17.17)	0.177 ^{***} (15.04)	0.182 ^{***} (16.55)	0.203 ^{***} (15.01)	0.194 ^{***} (16.10)	
$Debt_{i,t}$	-0.034 ^{****} (-11.66)	-0.056 ^{***} (-12.15)	-0.033 ^{***} (-17.20)	-0.067*** (-15.23)	-0.062*** (-13.70)	-0.038 ^{***} (-16.07)	-0.049 ^{***} (-11.22)	-0.071 ^{****} (-7.20)	-0.316 ^{***} (-14.41)	-0.034 ^{***} (-13.77)	-0.319 ^{***} (-12.55)	-0.031 ^{***} (-13.50)	
$Stk_{i,t}$	0.006^{*} (1.29)	0.009^{*} (1.73)	0.003 (0.54)	-0.004 (-0.73)	0.030 ^(4.39)	0.031 (0.26)	0.008*** (5.42)	0.019*** (4.23)	0.009*** (12.31)	0.001 (11.55)	0.005^{***} (4.65)	0.011** (2.46)	
$Size_{i,t}$	-0.010 (-0.84)	-0.013 (-0.05)	0.110 (0.24)	-0.018 ^{**} (-2.61)	-0.011 ^{**} (-2.51)	-0.043 [*] (-1.65)	-0.050 ^{**} (-2.24)	-0.018 ^{**} (-2.20)	0.015 ^{***} (7.34)	0.020^{**} (2.78)	0.008^{***} (4.86)	-0.026 (-0.52)	
Cons	-0.044 ^{***} (-10.20)	-0.105 ^{***} (-13.18)	-0.014 ^{***} (-15.41)	-0.101 ^{***} (-11.18)	0.072 ^{***} (5.04)	-0.163 [*] (-0.93)	0.075^{*} (1.82)	0.060^{***} (6.44)	-0.134 [*] (-1.00)	-0.027 ^{***} (-6.10)	-0.109 ^{***} (-11.24)	0.039 ^{***} (7.77)	
Firm	control	control	control	control	control	control	control	control	control	control	control	control	
Year	control	control	control	control	control	control	control	control	control	control	control	control	
Sample size	2142	2142	2142	2142	2142	2142	2142	2142	2142	2142	2142	2142	
Arellano–Bond test (m1)	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	
Arellano–Bond test (m2)	0.551	0.516	0.480	0.422	0.399	0.544	0.489	0.422	0.524	0.408	0.387	0.460	
Sargan test	0.804	0.750	0.799	0.825	0.711	0.810	0.891	0.806	0.658	0.670	0.755	0.801	

Table 2. Financial constraints vs. innovation efficiency on R&D protection with proceeds from operating assets sales

Notes: This table reports the two-step GMM estimation results using unbalanced panel data. The dependent variable is R&D investment (Rd), while *Cons* represents the constant. Values in parentheses are standard errors corrected by Windmeijer (2005). ***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively.

		Invention	(IE = IE')			Utility mode	el ($\overline{IE} = \overline{IE''}$)	Design $(IE = IE^{''})$				
Variable	Eq. (1)	Eq. (5)	Eq. (6)	Eq. (7)	Eq. (1)	Eq. (5)	Eq. (6)	Eq. (7)	Eq. (1)	Eq. (5)	Eq. (6)	Eq. (7)	
$Rd_{i,t-1}$	0.584 ^{***} (92.33)	0.602 ^{***} (110.21)	0.499 ^{***} (82.89)	0.510 ^{***} (103.06)	0.572 ^{***} (78.86)	0.561 ^{***} (79.99)	0.471 ^{***} (84.39)	0.501 ^{***} (80.06)	0.572 ^{***} (90.12)	0.569 ^{***} (82.12)	0.505 ^{***} (87.30)	0.543 ^{***} (99.06)	
$Rd_{i,t-1}^2$	-0.054 ^{***} (-11.35)	-0.048 ^{***} (-17.33)	-0.051 ^{***} (-18.02)	-0.045 ^{***} (-14.35)	-0.062 ^{***} (-13.33)	-0.039 ^{***} (-9.44)	-0.022 ^{***} (-8.35)	-0.051 ^{***} (-15.02)	-0.019* ^{***} (-12.88)	-0.022 ^{***} (-10.02)	-0.012 ^{***} (-8.33)	-0.034 ^{***} (-10.01)	
$Inv_{i,t}$	0.019 ^{****} (4.53)	0.37 (0.42)	0.027^{**} (1.99)	-0.031 (-0.95)	0.033 ^{****} (3.81)	0.004^{*} (1.45)	0.015 ^{***} (4.12)	-0.022 (-1.34)	0.002 [*] (1.44)	-0.072 (-0.33)	0.055^{*} (1.48)	0.011 (0.15)	
$Inv_{i,t} \times FC_{i,t-1}$		-0.011 (-6.03)		0.047 (0.40)		-0.070 (-2.41)		0.077 (0.60)		-0.053 (-0.16)		-0.002 (-3.88)	
$Inv_{i,t} \times IE_{i,t-1}$			0.048 ^{***} (5.17)	0.001 ^{***}			0.102 ^{**} (2.76)	0.000****			-0.117 (-0.84)		
$Invi, t \times FC_{i,t-1} \times IE_{i,t-1}$				0.091 (4.50)				0.008 (3.99)				0.277 (0.34)	
$FC_{i,t-1}$	-0.093 ^{****} (-22.16)	-0.084 ^{***} (-19.17)	-0.077 ^{****} (-16.59)	-0.065 (-16.75)	-0.076 ^{**} (-16.33)	-0.079 ^{***} (-14.33)	-0.061 ^{***} (-13.77)	-0.074 (-12.36)	-0.065 ^{***} (-13.77)	-0.091 ^{***} (-14.56)	-0.088 ^{***} (-12.78)	-0.059 (-13.05)	
$IE_{i,t-1}$	0.282^{+++}	0.310^{-10}	0.244^{+++} (13.45)	0.275^{-10}	0.441 (11.32)	0.398^{+++} (18.22)	0.402^{+++}	$0.455^{(10,25)}$	0.206 (16.10)	0.297^{+++} (13.17)	0.303^{+++}	0.311 (15.94)	
$Growth_{i,t}$	0.032 ^{***} (12.11)	0.100 ^{***} (9.34)	0.044 ^{***} (8.20)	0.056 ^{***} (9.77)	0.071**** (13.11)	0.079 ^{***} (12.21)	0.065^{***} (8.11)	0.051^{****} (8.10)	0.049 ^{***} (9.12)	0.050 ^{***} (7.06)	0.081^{***} (8.12)	0.062 ^{***} (7.29)	
$Cf_{i,t}$	0.202 ^{***} (34.15)	0.218 ^{***} (30.11)	0.199 ^{***} (27.92)	0.202 ^{***} (24.33)	0.207 ^{**} (21.86)	0.161 ^{***} (20.08)	0.211 ^{***} (22.18)	0.230 ^{***} (21.10)	0.173 ^{***} (20.28)	0.185 ^{***} (18.71)	0.206 ^{***} (16.67)	0.200 ^{***} (14.32)	
$Debt_{i,t}$	-0.090 (-9.16)	-0.077 (-10.09)	-0.057 (-9.08)	-0.062 (-10.12)	-0.076 (-6.11)	-0.070 (-11.46)	-0.049 (-10.21)	-0.053 (-9.20)	-0.81 (-10.05)	-0.086 (-7.84)	-0.045 (-8.19)	-0.051 (-7.06)	
$Stk_{i,t}$	0.044 ^{***} (5.01)	0.051 ^{***} (4.10)	0.034 ^{**} (2.55)	0.028 ^{***} (3.16)	0.124 [*] (1.95)	0.101 ^{****} (3.69)	0.022 ^{***} (5.04)	0.031 [*] (1.99)	0.111 (0.56)	0.050 ^{***} (3.98)	0.045 ^{***} (3.36)	0.060 ^{***} (5.16)	
$Size_{i,t}$	-0.007 (-2.78)	-0.016 (-1.52)	-0.025 (1.97)	-0.043 (-1.99)	-0.029	-0.033 (-2.83)	0.028 (0.56)	0.046 (1.43)	-0.033 (-1.35)	0.041 (1.82)	0.054 (1.35)	0.101 (0.56)	
Cons	0.013 ^{***} (5.42)	0.088 ^{***} (7.41)	-0.042 ^{***} (-5.50)	-0.036 ^{***} (-5.07)	0.120 ^{***} (8.11)	-0.050 ^{***} (-3.22)	-0.122 [*] (-1.43)	-0.105 ^{***} (-3.11)	0.220 ^{***} (3.49)	-0.076 ^{****} (-10.11)	-0.107 ^{***} (-7.21)	0.050 ^{***} (4.77)	
Firm	control	control	control	control	control	control	control	control	control	control	control	control	
Year	control	control	control	control	control	control	control	control	control	control	control	control	
Sample size	2142	2142	2142	2142	2142	2142	2142	2142	2142	2142	2142	2142	
Arellano–Bond test (m1)	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	
Sargan test (m2)	0.487 0.766	0.592	0.379	0.506	0.452	0.511	0.582	0.455 0.792	0.512	0.504 0.677	0.401	0.588	

Table 3. Financial constraints vs. innovation efficiency on R&D protection with proceeds from financial assets sales

Notes: This table reports the two-step GMM estimation results using unbalanced panel data. The dependent variable is R&D investment (*Rd*), while Cons represents the constant. Values in parentheses are standard errors corrected by Windmeijer (2005). ***, **, and * denote significance at 1%, 5%, and 10% levels respectively.

5.4 Robustness test

For prudence sake, this section replaces the financial constraints index, proposed by Howell(2016), with the well-known *KZ* index (Kaplan and Zingales, 1997) to recheck the robustness of the findings. The robustness test results are reported in Tables A.2 and A.3 in the Appendix. The results are generally consistent with the main findings aforementioned.

As shown by Tables A.2 and A.3, R&D-intensive firms in general are inclined to protect their R&D investments by selling operating assets and financial assets, except the case of design patent with financial asset sales. R&D protection with asset sale is more pronounced for firms with high innovation efficiency, especially in the case of operating assets sales. Meanwhile, innovation efficiency has positive moderating effects on the R&D protection with asset sales while financial constraint negatively moderates the relationship between R&D investment and asset sales in the cases of invention and utility model. Furthermore, finically constrained firms with high innovation efficiency in utility model patent are found to protect their R&D investment with operating assets sale, and those with high innovation efficiency in utility model patent.

Tables A.2 and A.3 also indicate some anomalies induced by the incorporation of *KZ* index. The coefficients on the interaction terms of financial constraint and asset sales (namely, $KZ \times OP$, $KZ \times Inv$) in Equation (5) are both significantly positive in the case of design patent. Similarly, the coefficients on financial constraint (*KZ*) and its interaction terms with asset sales (*OP* and *Inv*) in Equation (7) are positive for the case of design patent, while the coefficients in other cases are significantly negative. These anomalies suggest that financially constrained firms would like to increase rather than reduce R&D investment in the design patent case, which is contrary to the findings in other cases with innovative patents.

However, these anomalies are against the rationale that firms are less likely to protect superficial innovation rather than breakthrough innovation with asset sales, especially when they are financially constrained. It is worth noting that these anomalies should not affect the main findings of this paper, since it mainly focuses on cutting-edge innovations, namely innovation and utility model patents, rather than design patent with superficial novelty.

6. Conclusion

This study explores the determinants of R&D smoothing with financing asset sales from the perspectives of financial constraints and innovation efficiency. Specifically, this study examines the countervailing effects of financial constraints and innovation efficiency on R&D smoothing with financing asset sales using evidence from China's R&D-intensive firms spanning 2009 to 2016. The study considers three types of innovation efficiencies and two financial constraint indicators to ensure the broadest possible coverage. The empirical results obtained by the two-step GMM estimation are generally consistent and offer interesting findings regarding R&D investment activities.

R&D-intensive firms in China are generally found to protect their R&D investments by selling operating assets and financial assets. Meanwhile, financial constraints are found to have negative moderating effects on the relationship between R&D investments and financing asset sales, while innovation efficiency is found to positively moderate the latter. More interestingly, financially constrained firms with high innovation efficiency are found to increase their R&D investments with financing asset sales. These findings show that firms with high innovation efficiency are more likely to increase their R&D investments by selling assets, while those with high financial constraints are inclined to reduce their R&D investments unless they have high innovation efficiency. These findings are more pronounced for operating assets sales than for financial asset sales. The behavior of selling asset to protect R&D is more pronounced in firms with high efficiency in core innovation, namely, invention patents and utility model patents. These findings suggest that financially constrained firms do not adopt the R&D smoothing strategy with financing asset sales unless they possess high innovation efficiency. These firms are largely driven by the financing motivation to sell assets to satisfy needs more urgent than that of protecting R&D investments. By contrast, firms with high innovation efficiency, even when financially constrained, are mainly motivated by the efficiency incentive to increase their R&D investments with financing asset sales. As a result, prior R&D performance is a major driver for R&D smoothing with proceeds from asset sales, though financial constraints are an alternative motivation for financing through assets sales.

Overall, these findings further our understanding of the practice of R&D smoothing with financing asset sales by clarifying the roles of innovation efficiency and financial constraints. For one thing, the findings complement previous studies regarding R&D persistence, such as Ganter and Hecker (2013), Kang et al.(2017) as well as Coad (2019), by highlighting the opposite effects of innovation efficiency and financial constraints in R&D smoothing. For another, the findings provide new insights into asset divesture theory, including Arnold et al. (2018)and Edmans and Mann (2019), by identifying two under-researched motivations for financing asset sales. This study is expected to build a bridge between R&D investment management and asset divesture from the perspective of R&D smoothing with financing asset sales.

Our findings also have the following policy implications. Firstly, R&D-intensive firms should preserve financial flexibility in corporate liquidity management, since they are easily subject to financial constraints. Financial flexibility enables the firms to alleviate financial distress at a low cost when facing negative shocks. Financially flexible firms can keep their R&D expenditures relatively smooth, and therefore, avoid the very large adjustment costs of discontinued or suspended R&D investments. Secondly, R&D-intensive firms should consider the R&D smoothing strategy with financing asset sales if they are financially constrained. Asset sales are an alternative financing source for R&D investments in emerging markets with limited external financing opportunities like China. Although the proceeds may be limited, they are especially valuable as a way for R&D-intensive firms to protect their value-enhancing R&D investments without having to resort to more costly external financing. Finally, R&D-intensive firms should pay attention to their innovation efficiency when smoothing their R&D investments through financing asset sales. Innovation efficiency not only provides a strong motivation to sell assets for R&D smoothing, but also determines the innovation outcomes obtained from the R&D smoothing strategy. To a large extent, the performance of the R&D smoothing strategy with financing asset sales would depend on the firms' innovation efficiencies, especially for those who are financially constrained.

A limitation of this study may arise from its focus on the listed R&D-intensive firms in China, since other firms may also take part in R&D investments and asset sales. The primary reason is that the R&D-intensive firms have a more pronounced R&D orientation and a stronger incentive to smooth their R&D investments. Even though these firms publicly listed, they are easily susceptible to financial constraints due to the unique characteristics of R&D investments, especially in a typical emerging market like China. Financing asset sales provides these firms an alternative to protect their R&D investment if they are financially constrained. The research findings are expected to be further verified in more comprehensive studies. Moreover, the impact of R&D smoothing strategy with financing asset sales on firm performance deserves further study, which is beyond the scope this paper.

Journal Proposi

References

- Aboody, D., & Lev, B. (2000). Information asymmetry, R&D, and insider gains. *Journal of Finance*, 55(6), 2747-2766.
- Ahrends, M., Drobetzb, W., & Puhan, T. X. (2018). Cyclicality of growth opportunities and the value of cash holdings. *Journal of Financial Stability*, 37, 74-96.
- Almeida, H., & Campello, M. (2007). Financial constraints, asset tangibility, and corporate investment. *Review of Financial Studies*, 20(5), 1429-1460.
- Arellano, M., & Bover, O. (1995). Another look at the instrumental variable estimation of error-components models. *Journal of Econometrics*, 68(1), 29-51.
- Arnold, M., Hackbarth, D., & Puhan, T. X. (2018). Financing asset sales and business cycles. *Review of Finance*, 243-277.
- Balachandra, R., Brockhoff, K. K., & Pearson, A. W. (1996). R&D project termination decisions: Processes, communication, and personnel changes. *Journal of Product Innovation Management*, 13(3), 245-256.
- Balakrishnan, K., Billings , M., Kelly, B., & Ljungqvist , A. (2014). Shaping liquidity: On the causal effects of voluntary disclosure. *Journal of Finance*, 69(5), 2237-2278.
- Bandyopadhyay, A., & Barua, N. M. (2016). Factors determining capital structure and corporate performance in India: Studying the business cycle effects. *Quarterly Review of Economics* and Finance, 61, 160-172.
- Banerjee, T., & Siebert, R. (2017). Dynamic impact of uncertainty on R&D cooperation formation and research performance: Evidence from the bio-pharmaceutical industry. *Research Policy*, 46, 1255-1271.
- Beck, T., Demirgüç-Kunt, A., Laeven, L., & Maksimovic, V. (2006). The determinants of financing obstacles. *Journal of International Money and Finance*, 25, 932-952.
- Benmelech, E., & Bergman, N. K. (2009). Collateral pricing. *Journal of Financial Economics*, 91(3), 339-360.
- Berger, A. N., & Udell, G. F. (1990). Collateral, loan quality, and bank risk. *Journal of Monetary Economics*, 25, 21-42.
- Bernanke, B., Gertler, M., & Gilchrist, S. (1996). The financial accelerator and the flight to quality. *Review of Economics and Statistics*, 78(1), 1-15.
- Bernini, M., & Montagnoli, A. (2017). Competition and financial constraints: A two-sided story. *Journal of International Money and Finance*, 70, 88-109.
- Bessler, W., Drobetz, W., Haller, R., & Meier, I. (2013). The international zero-leverage phenomenon. *Journal of Corporate Finance*, 23, 196-221.
- Bhagat, S., & Welch, I. (1995). Corporate research & development investments international comparisons. *Journal of Accounting and Economics*, *19*(2-3), 443-470.
- Bond, S., & Meghir, C. (1994). Dynamic investment models and the firm's financial policy. *Review of Economic Studies*, 61(2), 197-222.
- Bond, S., Elston, J. A., Mairesse, J., & Mulkay, B. (2003). Financial factors and investment in Belgium, France, Germany, and the United Kingdom: A comparison using company panel data. *Review of Economics and Statistics*, 85(1), 153-165.
- Borisova, G., & Brown, J. R. (2013). R&D sensitivity to asset sale proceeds: New evidence on financing constraints and intangible investment. *Journal of Banking and Finance, 37*, 159-173.

- Borisova, G., John, K., & Salotti, V. (2013). The value of financing through cross-border asset sales: Shareholder returns and liquidity. *Journal of Corporate Finance*, *22*, 320-344.
- Brown, J. R., & Petersen, B. C. (2011). Cash holdings and R&D smoothing. *Journal of Corporate Finance*, *17*, 694-709.
- Brown, J. R., & Petersen, B. C. (2015). Which investments do firms protect? Liquidity management and real adjustments when access to finance falls sharply. *Journal of Financial Intermediation*, 24, 441-465.
- Brown, J. R., Fazzari, S. M., & Petersen, B. C. (2009). Financing innovation and growth: Cash flow, external equity, and the 1990s R&D boom. *Journal of Finance*, 64(1), 151-185.
- Brown, J. R., Martinsson, G., & Petersen, B. C. (2012). Do financing constraints matter for R&D? *European Economic Review*, 56, 1512-1529.
- Campello, M., Graham, J. R., & Harvey, C. R. (2010). The real effects of financial constraints: Evidence from a financial crisis. *Journal of Financial Economics*, 97(3), 470-487.
- Chay, J. B., Park, S. H., Kim, S., & Suh, J. (2015). Financing hierarchy: Evidence from quantile regression. *Journal of Corporate Finance*, *33*, 147-163.
- Chen, H., & Chen, S. (2012). Investment-cashflow sensitivity cannot be a good measure of financial constraints: Evidence from the time series. *Journal of Financial Economics*, 103(2), 393-410.
- Chen, Y., Hua, X., & Boateng, A. (2017). Effects of foreign acquisitions on financial constraints, productivity and investment in R&D of target firms in China. *International Business Review*, 26, 640-651.
- Christodoulou, D., Lev, B., & Ma, L. (2018). The productivity of Chinese patents: The role of business area and ownership type. *International Journal of Production Economics*, 199, 107-124.
- Coad, A. (2019). Persistent heterogeneity of R&D intensities within sectors: Evidence and policy implications. *Research Policy*, 48, 37-50.
- Cruz-Cázares, C., Bayona-Sáez, C., & García-Marco, T. (2013). You can't manage right what you can't measure well: Technological innovation efficiency. *Research Policy*, *42*, 1239-1250.
- Dang, V. A. (2013). An empirical analysis of zero-leverage firms: New evidence from the UK. *International Review of Financial Analysis*, *30*, 189-202.
- David, P. A., Hall, B. H., & Toole, A. A. (2000). Is public R&D a complement or substitute for private R&D? A review of the econometric evidence. *Research Policy*, *29*, 497-529.
- Desai, C. A., & Gupta, M. (2019). Size of financing need and the choice between asset sales and security issuances. *Financial Management*, 48(2), 677-718.
- Dierickx, I., & Cool, K. (1989). Asset stock accumulation and competitive advantage. *Management Science*, 35(12), 1504-1511.
- Demir, F. (2009). Financial liberalization, private investment and portfolio choice: Financialization of real sectors in emerging markets. *Journal of Development Economics*, 88 (2):314-324.
- Drivera, C., & Guedes, M. J. (2012). Research and development, cash flow, agency and governance: UK large companies. *Research Policy*, *41*, 1565-1577.
- Dutta, S., & Nezlobin, A. (2017). Information disclosure, firm growth, and the cost of capital. *Journal of Financial Economics*, 123(2), 415-431.
- Edmans, A., & Mann, W. (2019). Financing through asset sales. Management Science, 65(7),

3043-3060.

- Erel, I., Jang, Y., & Weisbach, M. S. (2015). Do acquisitions relieve target firms' financial constraints? *Journal of Finance*, 70(1), 289-328.
- Fazzari, S. M., Hubbard, R. G., Petersen, B. C., Blinder, A. S., & Poterba, J. M. (1988). Financing constraints and corporate investment. *Brookings Papers on Economic Activity*, 1988(1), 141-206.
- Fee, C. E., Hadlock, C. J., & Pierce, J. R. (2009). Investment, financing constraints, and internal capital markets: Evidence from the advertising expenditures of multinational firms. *Review of Financial Studies*, 22(6), 2361-2392.
- Fernandez, V. (2017). The finance of innovation in Latin America. *International Review of Financial Analysis*, 53, 37-47.
- Foley-Fisher, N., Ramcharan, R., & Yu, E. (2016). The impact of unconventional monetary policy on firm financing constraints: Evidence from the maturity extension program. *Journal of Financial Economics*, 122(2), 409-429.
- Fu, X. (2012). How does openness affect the importance of incentives for innovation? *Research Policy*, *41*, 512-523.
- Ganter, A., & Hecker, A. (2013). Persistence of innovation: Discriminating between types of innovation and sources of state dependence. *Research Policy*, *42*, 1431-1445.
- Gao, W., & Chou, J. (2015). Innovation efficiency, global diversification, and firm value. *Journal* of Corporate Finance, 30, 278-298.
- Ghosal, V., & Ye, Y. (2015). Uncertainty and the employment dynamics of small and large businesses. *Small Business Economics*, 44(3), 529-558.
- Griffin, P. A., Hong, H. A., & Ryou, J. W. (2018). Corporate innovative efficiency: Evidence of effects on credit ratings. *Journal of Corporate Finance*, 51, 352-373.
- Grossman, G. M., & Helpman, E. (1991). *Innovation and growth in the global economy*. Cambridge: MIT press.
- Guan, J., & Chen, K. (2012). Modeling the relative efficiency of national innovation systems. *Research Policy*, 41, 102-115.
- Guariglia, A., & Liu, P. (2014). To what extent do financing constraints affect Chinese firms' innovation activities? *International Review of Financial Analysis*, *36*, 223-240.
- Guney, Y., Karpuz, A., & Ozkan, N. (2017). R&D investments and credit lines. Journal of Corporate Finance, 46, 261-283.
- Hadlock, C. J., & Pierce, J. R. (2010). New evidence on measuring financial constraints: Moving beyond the KZ index. *Review of Financial Studies*, 23(5), 1909-1940.
- Hahn, K. (2019). Innovation in times of financialization: Do future-oriented innovation strategies suffer? Examples from German industry. *Research Policy*, *48* (4):923-935.
- Hall, B. H. (2002). The financing of research and development. *Oxford Review of Economic Policy*, *18*(1), 35-51.
- Hall, B. H., & Lerner, J. (2010). The financing of R&D and innovation. In B. H. Hall, & N. Rosenberg, *Handbook of the Economics of Innovation* (pp. 609-639). Amsterdam: North-Holland Elsevier Publishers.
- He, Z., & Wintoki, M. B. (2016). The cost of innovation: R&D and high cash holdings in U.S. frims. *Journal of Corporate Finance*, *41*, 280-303.
- Himmelberg, C. P., & Petersen, B. C. (1994). R&D and internal finance: A panel study of small

firms in high-tech industries. Review of Economics and Statistics, 76(1), 38-51.

- Hirshleifer, D., Hsu, P.-H., & Li, D. (2013). Innovative efficiency and stock returns. *Journal of Financial Economics*, 107(3), 632-654.
- Hoegl, M., Gibbert, M., & Mazursky, D. (2008). Financial constraints in innovation projects: When is less more? *Research Policy*, 37, 1382-1391.
- Hottenrott, H., & Peters, B. (2012). Innovative capability and financing constraints for innovation: More money, more innovation? *Review of Economics and Statistics*, 94(4), 1126-1142.
- Hovakimian, G., & Titman, S. (2006). Corporate investment with financial constraints: Sensitivity of investment to funds from voluntary asset sales. *Journal of Money, Credit and Banking, 38*(2), 357-374.
- Howell, A. (2016). Firm R&D, innovation and easing financial constraints in China: Does corporate tax reform matter? *Research Policy*, 45, 1996-2007.
- Hyytinen, A., & Pajarinen, M. (2008). Opacity of young businesses: Evidence from rating disagreements. *Journal of Banking and Finance*, *32*, 1234-1241.
- Jiang, X., & Yuan, Q. (2018). Institutional investors' corporate site visits and corporate innovation. *Journal of Corporate Finance, 48*, 148-168.
- Kang, T., Baek, C., & Lee, J.-D. (2017). The persistency and volatility of the firm R&D investment: Revisited from the perspective of technological capability. *Research Policy*, 46, 1570-1579.
- Kamien, M., and N. Schwartz. (1978). Self-financing of an R&D project. *American Economic Review*, 68:252-261.
- Kaplan, S. N., & Zingales, L. (1997). Do investment-cash flow sensitivities provide useful measures of financing constraints? *Quarterly Journal of Economics*, 112(1), 169-215.
- Khatami, S. H., Marchica, M.-T., & Mura, R. (2015). Corporate acquisitions and financial constraints. *International Review of Financial Analysis*, 40, 107-121.
- Knut, B., Petersen, S. S., & Riillo, C. A. (2017). The impact of standards and regulation on innovation in uncertain markets. *Research Policy*, 46, 249-264.
- Lach, S., & Schankerman, M. (1989). Dynamics of R&D and investment in the scientific sector. *Journal of Political Economy*, 97(4), 880-904.
- Li, X. (2009). China's regional innovation capacity in transition: An empirical approach. *Research Policy*, *38*, 338-357.
- Orhangazi, O. (2008). Financialisation and capital accumulation in the non-financial corporate sector: A theoretical and empirical investigation on the US economy: 1973-2003. *Cambridge Journal of Economics, 32* (6):863-886.
- Máñez, J. A., Rochina-Barrachina, M. E., Sanchis, A., & Sanchis, J. A. (2009). The role of sunk costs in the decision to invest in R&D. *Journal of Industrial Economics*, *57*(4), 712-735.
- Mann, W. (2018). Creditor rights and innovation: Evidence from patent collateral. *Journal of Financial Economics*, *130*(1), 25-47.
- Mohamed, A., & Schwienbacher, A. (2016). Voluntary disclosure of corporate venture capital investments. *Journal of Banking and Finance*, 68, 69-83.
- Mohamed, E. B., & Shehata, M. A. (2017). R&D investment–cash flow sensitivity under managerial optimism. *Journal of Behavioral and Experimental Finance*, 14, 1-4.
- Moshirian, F., Nanda, V., Vadilyev, A., & Zhang, B. (2017). What drives investment-cash flow sensitivity around the World? An asset tangibility perspective. *Journal of Banking and*

Finance, 77, 1-17.

- Mulier, K., Schoors, K., & Merlevede, B. (2016). Investment-cash flow sensitivity and financial constraints: Evidence from unquoted European SMEs. *Journal of Banking and Finance*, 73, 182-197.
- Pellegrinoa, G., & Savona, M. (2017). No money, no honey? Financial versus knowledge and demand constraints on innovation. *Research Policy*, *46*, 510-521.
- Peter, R. H., & Taylor, L. A. (2017). Intangible capital and the investment-q relation. *Journal of Financial Economics*, 123(2), 251-272.
- Pindyck, R. S. (1993a). Investments of uncertain cost. *Journal of Financial Economics*, 34(1), 53-76.
- Pindyck, R. S. (1993b). A note on competitive investment under uncertainty. *American Economic Review*, 83(1), 273-277.
- Rauh, J. D. (2006). Investment and financing constraints: Evidence from the funding of corporate pension plan. *Journal of Finance*, *61*(1), 33-71.
- Richard, B., & Bond, S. (1998). Initial conditions and moment restrictions in dynamic panel data models. *Journal of Econometrics*, 87(1), 115-143.
- Sasidharan, S., Jijo Lukose, P. J., & Komera, S. (2015). Financing constraints and investments in R&D: Evidence from Indian manufacturing firms. *Quarterly Review of Economics and Finance*, 55, 28-39.
- Schiantarelli, F. (1996). Financial constraints and investment: Methodological issues and international evidence. *Oxford Review of Economic Policy*, *12*(2), 70-89.
- Sears, J. B. (2017). When are acquired technological capabilities complements rather than substitutes? A study on value creation. *Journal of Business Research*, 78, 33-42.
- Sears, J. B. (2018). Post-acquisition integrative versus independent innovation: A story of dueling success factors. *Research Policy*, 47(9), 1688-1699.
- Seru, A. (2014). Firm boundaries matter: Evidence from conglomerates and R&D activity. *Journal* of Financial Economics, 111(2), 381-405.
- Shao, Y., & Xiao, C. (2019). Corporate tax policy and heterogeneous firm innovation: Evidence from a developing country. *Journal of Comparative Economics*, 47, 470-486.
- Shleifer, A., & Vishny, R. W. (1992). Liquidation values and debt capacity: A market equilibrium approach. *Journal of Finance*, 47(4), 1343-1366.
- Solow, R. M. (1957). Technical change and the aggregate production function. *Review of Economics and Statistics*, 39(3), 312-320.
- Stephen, B., Elston, J. A., Mairesse, J., & Mulkay, B. (2003). Financial factors and investment in Belgium, France, Germany, and the United Kingdom: A comparison using company panel data. *Review of Economics and Statistics*, 85(1), 153-165.
- Strebulaev, I. A., & Yang, B. (2013). The mystery of zero-leverage firms. Journal of Financial Economics, 109(1), 1-23.
- Ullah, S., Akhtar, P., & Zaefarian, G. (2018). Dealing with endogeneity bias: The generalized method of moments (GMM) for panel data. *Industrial Marketing Management*, 71, 69-78.
- Wang, E. C., & Huang, W. (2007). Relative efficiency of R&D activities: A cross-country study accounting for environmental factors in the DEA approach. *Research Policy*, 36, 260-273.
- Wang, Y. (2017). Debt market friction, firm-specific knowledge capital accumulation and macroeconomic implications. *Review of Economic Dynamics*, 26, 19-39.

- Wang, Y., Wei, Y., & Song, F. M. (2017). Uncertainty and corporate R&D investment: Evidence from Chinese listed firms. International Review of Economics and Finance, 47, 176-200.
- Whited, T. M., & Wu, G. (2006). Financial constraints risk. Review of Financial Studies, 19(2), 531-559.
- Windmeijer, F. (2005). A finite sample correction for the variance of linear efficient two-step GMM estimators. Journal of Econometrics, 126(1), 25-51.
- Zhang, W. (2015). R&D investment and distress risk. Journal of Empirical Finance, 32, 94-114.
- Zhang, S.C., & Zhang, B.T. (2016). The falling real investment puzzle: a view from financialization. Economic Research Journal (in Chinese), 51(12):32-46.
- Zheng, G., Wang, S., & Xu, Y. (2018). Monetary stimulation, bank relationship and innovation: Evidence from China. Journal of Banking and Finance, 89, 237-248.

Appendix



Figure A.1 Financial asset investment of firms in China

Notes: This figure shows the secular trend of the proportion of financial assets to total assets held by non-financial A-share listed companies in China from 2009 to 2019.

Jonuly

Variable	Definition	Description	Variable	Definition	Description
Rd	R&D intensity	R&D spending/Total assets	Age	Firm age	Natural logarithm of the years between establishment date (or merger date) and the sample period
Rd_{t-1}^2	Quadratic adjustment costs of R&D investment	Squared value of lagged R&D intensity	FC	Dummy variable of financial constraints	FC = 1 if the value of F in Model 5 is positive; otherwise, FC = 0.
ОР	Operating assets sales ratio	Net cash from disposing of fixed assets, intangible assets, and other long-term assets/Total assets	$IE^{'}$	Innovation efficiency of invention	Application of patents for invention /R&D spending
Inv	financial assets sale ratio	Net cash from disposing of investment assets /Total assets	IE"	Innovation efficiency of utility model	Application of patents of utility model /R&D spending
Cf	Cash flow ratio	Net cash flow from operating activities /Total assets	<i>IE</i> ‴	Innovation efficiency of design	Application of patents of designs/R&D spending
Growth	Investment opportunities	Average sales growth of last three years	θ	R&D smoothing degree by current operating assets sales	Absolute difference between the actual R&D investment of the firm and its theoretical value obtained in Equation (1), scaled by the standard deviation of those in the industry.
Stk	Net sock issue ratio	(Seasoned equity offerings + Rationed shares) /Total assets	θ'	R&D smoothing degree by financial assets sales	Absolute difference between the actual R&D investment of the firm and its theoretical value obtained in Equation (2), scaled by the standard deviation of those in the industry.
Debt	Net debt issue ratio	(Current debt - Lagged debt) /Total assets	ASale	Asset sales ratio	Sum of operating assets sale ratio and financial assets sale ratio
Size	Firm size	Natural logarithm of total assets	Lev	Level of debt	Total debt/Total assets
Ι	Investment spending	Cash paid to purchase fixed assets, intangible assets, and other long-term investments	Sales	Operating income	Sales from operating activities/Total assets

TableA.1. Variable definitions and description

	Invention $(IE = IE')$					Utility mode	el (IE = IE'')	Design(IE = IE''')				
Variable	Eq. (1)	Eq.(5)	Eq.(6)	Eq.(7)	Eq.(1)	Eq.(5)	Eq.(6)	Eq.(7)	Eq.(1)	Eq.(5)	Eq.(6)	Eq.(7)	
$Rd_{i,t-1}$	0.046 ^{***} (95.33)	0.002 ^{***} (94.20)	0.597 ^{***} (108.32)	0.599 ^{***} (99.25)	0.013 ^{***} (110.41)	0.029 ^{***} (97.32)	0.008^{***} (108.05)	0.595 ^{***} (89.39)	0.566 ^{***} (82.46)	0.547 ^{***} (79.05)	0.573 ^{***} (88.72)	0.068 ^{****} (75.30)	
$Rd_{i,t-1}^2$	-0.061 ^{***} (-17.33)	-0.055 ^{***} (-18.19)	-0.064 ^{***} (-18.11)	-0.059 ^{***} (-19.54)	-0.066 ^{***} (-23.33)	-0.050 ^{***} (-18.26)	-0.051 ^{***} (-21.44)	-0.048 ^{***} (-18.15)	-0.052 ^{***} (-10.33)	-0.044 ^{***} (-13.59)	-0.039 ^{***} (-12.35)	-0.042 ^{***} (-14.33)	
$OP_{i,t}$	0.040 ^{***} (7.25)	0.046^{*} (1.71)	0.039 ^{**} (2.44)	0.051 [*] (1.33)	0.029 ^{***} (6.40)	0.042 (1.32)	0.127 (0.55)	-0.011** (-2.62)	0.007^{***} (5.08)	0.011 [*] (1.95)	0.056 ^{**} (2.33)	-0.045* (-1.96)	
$OP_{i,t} \times KZ_{i,t-1}$		-0.077** (-2.94)	0.000***	-0.133 (-0.28)		-0.034 (-4.05)		-0.107* (-2.04)		0.086 (5.79)	0.075***	0.003 (2.91)	
$OP_{i,t} \times IE_{i,t-1}$			0.303 (4.97)	0.414			(10.36)	0.092***			0.075 (3.84)	0.102*	
$OP_{i,t} \times KZ_{i,t-1} \times IE_{i,t-1}$	0.045*	0.005	0.022*	(0.31)	0.000***	0.010*	0.010***	(4.20)	0.010*	0.000	0.025*	(1.16)	
$KZ_{i,t-1}$	-0.045 (-2.01)	-0.085 (-0.17)	-0.032 (-1.59)	-0.065 (-2.35)	-0.009 (-4.30)	(-1.33)	-0.018 (-3.79)	-0.026 (-2.36)	-0.012 (-1.75)	(0.56)	-0.025 (-2.01)	(1.47)	
$IE_{i,t-1}$	0.237 (13.31)	0.222 (12.80)	0.195 (10.24)	0.170 (9.05)	0.357 (9.48)	0.306 (8.33)	0.508 (11.12)	0.392 (8.30)	0.092 (5.29)	0.111 (9.53)	0.123 (12.54)	0.252 (9.74)	
$Growth_{i,t}$	0.035 (6.35)	0.104 (7.35)	0.077 (8.94)	0.073 (15.86)	0.044 (14.67)	0.081 (22.18)	0.028 (15.32)	0.056 (8.04)	0.040 (6.35)	0.100 (7.02)	0.075 (8.94)	0.036 (13.28)	
$Cf_{i,t}$	0.237 ^{***} (13.31)	0.173 ^{***} (9.04)	0.062 ^{***} (4.67)	0.061 ^{****} (8.24)	0.049 ^{***} (5.55)	0.138 ^{***} (7.43)	0.085 ^{***} (9.11)	0.028^{***} (14.78)	0.049 ^{***} (5.55)	0.029 ^{***} (6.77)	0.032 ^{***} (7.93)	0.038 ^{***} (7.23)	
$Debt_{i,t}$	-0.027*** (-19.20)	-0.022*** (-25.75)	-0.033** (-14.27)	-0.026 (-26.03)	-0.020*** (-21.06)	-0.031*** (-20.75)	-0.025*** (-13.05)	-0.019*** (-18.50)	-0.018*** (-21.06)	-0.028*** (-22.78)	0.022*** (-22.22)	-0.031*** (-18.06)	
$Stk_{i,t}$	0.083 (25.53)	0.077 (26.86)	0.079 (28.43)	0.065 (21.73)	0.022 [*] (1.93)	0.302 (0.37)	0.096 (0.33)	0.061 (5.85)	0.076^{*} (1.79)	0.036 ^{**} (2.47)	0.063 [*] (1.99)	0.051 (0.47)	
Size _{i,t}	-0.078 (-3.25)	-0.079 ⁺⁺ (-2.13)	-0.076 (-1.15)	-0.053 (-0.36)	-0.079 (-9.21)	-0.090 (-8.43)	-0.045 (-5.11)	-0.049 (-6.59)	-0.105 (-10.44)	-0.082 (-7.50)	-0.076 (-8.47)	0.880 (21.92)	
Cons	0.039 ^{***} (8.10)	0.036 ^{**} (2.64)	-0.018 [*] (-1.21)	-0.023* (-1.07)	0.021 (6.62)	-0.025* (-1.63)	-0.020 ^{***} (-5.08)	-0.031 (-8.61)	0.095 ^{***} (4.11)	-0.109*** (-6.59)	0.056 ^{***} (8.11)	-0.062^{+} (-1.18)	
Firm	control												
Year	control												
Sample size	2142	1933	2142	1933	2142	1933	2142	1933	2142	1933	2142	1933	
Arellano–Bond test (m1)	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	
Arellano–Bond test (m2)	0.300	0.296	0.288	0.305	0.287	0.301	0.308	0.299	0.332	0.351	0.295	0.344	
Sargan test	0.061	0.708	0.796	0.746	0.903	0.045	0.899	0.812	0.541	0.477	0.001	0.581	

Table A.2. Financial constraints (KZ) vs. innovation efficiency on R&D protection with proceeds from operating assets sales

Notes: This table reports the two-step GMM estimation results using unbalanced panel data. The dependent variable is R&D investment (Rd), while KZstands for financial constraints. Values in parentheses are the standard errors corrected by Windmeijer (2005). ***, **, and * denote significance at 1%, 5%, and 10% levels respectively.

	Invention $(IE = IE')$					Utility mode	I(IE = IE''))	$Design(IE = IE^{''})$				
Variables	Eq. (1)	Eq.(5)	Eq.(6)	Eq.(7)	Eq.(1)	Eq.(5)	Eq.(6)	Eq.(7)	Eq.(1)	Eq.(5)	Eq.(6)	Eq.(7)	
$Rd_{i,t-1}$	0.055 ^{***} (98.07)	0.588 ^{***} (86.97)	0.549 ^{***} (74.39)	0.582 ^{***} (69.82)	0.559 ^{***} (83.45)	0.572 ^{***} (93.58)	0.570 ^{***} (82.30)	0.040 ^{***} (105.10)	0.023 ^{***} (87.90)	0.578 ^{***} (74.38)	0.021 ^{****} (110.34)	0.019 ^{***} (73.66)	
$Rd_{i,t-1}^2$	-0.039 ^{***} (-15.10)	-0.036 ^{***} (-12.64)	-0.018 ^{***} (-12.21)	-0.023 ^{***} (-16.07)	-0.021 ^{***} (-11.62)	-0.025 ^{***} (-17.63)	-0.049 ^{***} (-14.06)	-0.056 ^{***} (-14.51)	-0.055 ^{***} (-18.02)	-0.057 ^{***} (-14.00)	-0.058 ^{***} (-21.57)	-0.061 ^{***} (-18.31)	
$Inv_{i,t}$	0.036 (3.73)	0.122 (1.48)	0.051^{+} (1.68)	-0.110 (-1.36)	0.018 (4.53)	-0.025 (-1.71)	0.034 (1.07)	-0.023 (-1.16)	0.206 (0.49)	0.019 (2.94)	0.020 ^{**} (2.37)	-0.041 (-1.59)	
$Inv_{i,t} \times KZ_{i,t-1}$		-0.082 (-3.02)		-0.052 (-4.10)		-0.112 (-2.23)		-0.048 (-2.71)		0.258 ⁺ (1.37)		0.054° (1.80)	
$Inv_{i,t} \times IE_{i,t-1}$			0.208 ^{***} (4.93)				0.059 ^{***} (4.16)				0.263 ^{**} (2.02)		
$Inv_{i,t} \times KZ_{i,t-1} \times IE_{i,t-1}$				0.022 ^{***} (4.07)				0.151 ^{**} (2.32)				0.120 (0.35)	
$KZ_{i,t-1}$	-0.032** (-2.38)	-0.016 [*] (-1.55)	-0.053 [*] (-1.99)	-0.036 ^{***} (-5.47)	-0.022 [*] (-1.93)	0.002 (0.37)	-0.096 (-0.33)	0.51 (0.85)	-0.045 ^{**} (-2.79)	-0.039 ^{***} (-10.19)	-0.041 ^{****} (-5.54)	0.079 [*] (1.26)	
$IE_{i,t-1}$	0.301 ^{***} (26.15)	0.155 ^{***} (10.13)	0.173 ^{***} (9.04)	0.062 ^{***} (4.67)	0.361**** (8.24)	0.049 ^{****} (5.55)	0.208 ^{***} (7.43)	0.385 ^{***} (11.11)	0.028 ^{***} (14.78)	0.057 ^{***} (11.48)	0.146 ^{***} (6.06)	0.173 ^{****} (9.04)	
$Growth_{i,t}$	0.078^{***} (10.02)	0.059 ^{***} (8.90)	0.057 ^{***} (7.51)	0.033**** (12.10)	0.074*** (13.20)	0.052^{***} (9.00)	0.081 ^{***} (13.58)	0.055 ^{***} (8.56)	0.049^{***} (12.01)	0.028^{***} (11.00)	0.037 ^{***} (10.29)	0.077 ^{***} (18.25)	
$Cf_{i,t}$	0.136 ^{***} (21.84)	0.116 ^{***} (11.00)	0.164 ^{****} (20.29)	0.125 ^{***} (12.22)	0.220 ^{**} (12.24)	0.121 ^{**} (13.43)	0.111 (14.51)	0.233 ^{***} (13.29)	0.228 ^{***} (15.42)	0.012 (11.18)	0.062^{***} (9.31)	0.052^{***} (3.60)	
$Debt_{i,t}$	-0.236 ^{***} (-24.97)	-0.130 ^{***} (-20.55)	-0.229 ^{***} (-31.23)	-0.132*** (-18.72)	-0.204 ^{***} (-16.10)	-0.117 ^{***} (-27.55)	-0.203 ^{***} (-21.06)	-0.109 ^{***} (-30.49)	-0.134 ^{***} (-15.00)	-0.047 ^{***} (-16.06)	-0.101 ^{***} (-11.24)	0.059 ^{****} (12.72)	
$Stk_{i,t}$	0.124 ^{****} (5.43)	0.033 ^{***} (3.42)	0.114** (2.41)	0.098 ^{**} (2.75)	0.081*(1.04)	0.073** (2.07)	0.065 ^{***} (5.82)	0.083*(1.34)	0.134 (0.57)	0.204 (0.46)	0.109*(1.24)	0.169 ^{**} (2.72)	
$Size_{i,t}$	-0.125 [*] (-1.59)	-0.051 ^{***} (-2.50)	0.129 [*] (1.52)	-0.075 (-0.28)	-0.006 ^{***} (-4.07)	-0.096 [*] (-1.41)	-0.047 ^{**} (-2.55)	0.010 [*] (1.06)	0.104 ^{**} (2.49)	-0.231 (-0.30)	-0.152 ^{***} (-2.13)	0.235 [*] (1.16)	
Cons	-0.066 ^{**} (-2.63)	0.113^{*} (1.05)	-0.075 (-0.33)	-0.004 (-0.35)	0.164 ^{***} (11.92)	0.057^{*} (1.97)	-0.050 ^{***} (-8.43)	0.019^{***} (5.43)	-0.141 [*] (-1.59)	0.023 ^{***} (3.27)	-0.157 ^{***} (-7.18)	0.224^{***} (4.79)	
Firm	control												
Year	control												
Sample size	2142	1933	2142	1933	2142	1933	2142	1933	2142	1933	2142	1933	
Arellano–Bond test (m1)	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	
Arellano–Bond test (m2)	0.384	0.297	0.420	0.286	0.346	0.405	0.391	0.299	0.394	0.308	0.339	0.385	
Sargan test	0.881	0.534	0.868	0.611	0.076	0.910	0.885	0.906	0.617	0.692	0.528	0.604	

Table A.3. Financial constraints (KZ) vs. innovation efficiency on R&D protection with the proceeds from financial asset sales

Notes: This table reports the two-step GMM estimation results using unbalanced panel data. The dependent variable is R&D investment (Rd), while KZstands for financial constraints. Values in parentheses are the standard errors corrected by Windmeijer (2005). ***, **, and * denote significance at 1%, 5%, and 10% levels respectively.

Highlights

- We examine the determinants of R&D smoothing with financing asset sales
- R&D intensive firms are found to protect their R&D investments by selling assets •
- Financial constraints prohibit firms from adopting the R&D smoothing strategy
- Firms with high innovation efficiency are more inclined to adopt the strategy •
- ٠ Innovation efficiency is regarded as the major incentive for adopting the strategy

. fo

Author Statement

Title of Manuscript: <u>The determinants of R&D smoothing with asset sales: Evidence</u> <u>from R&D-intensive firms in China</u>

Corresponding Author Name/Email: <u>Hongbo HE/ hongbo_he@hnu.edu.cn</u>

In order to publish our paper in the *International Review of Economics and Finance* (IREF), all authors have read and signed the statements on (i) Author Agreement; (ii)Disclosure of Conflicts of Interest; (iii) Funding Source Declarations;(iv)Permission Notesand(v)Authorship Contributions

i. AuthorAgreement

We have seen and approved the final version of the manuscript being submitted. We warrant that the article is the authors' original work, hasn't received prior publication and isn't under consideration for publication elsewhere. We have followed ethics guidelines.

ii. Disclosure of Conflicts of Interest

We declare that we have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

iii.Funding Source Declarations

Duan LIU has obtained financial support from the National Natural Science Foundation of China (71972066), the Humanities and Social Science Foundation of the Ministry of Education of China (17YJA630056, 16YJA790058), and the Philosophy and Social Science Foundation of Hunan Province (18YBA085). Hongbo HE has obtained financial support from the National Natural Science Foundation of China (71573077, 71790593, 71521061, and 71673033).

Zhiyuan LI and Wenxuan HOU have no related interest with the article.

iv. Permission Notes

We confirm that the permission of the copyright owner(s) has been received to use anymaterial in the paper and is clearly identified and acknowledged in the text of the paper.

v. Authorship Contributions

- Conceived and designed the analysis: Hongbo HE;
- Collected the data: Zhiyuan LI
- Performed the analysis : Duan LIU
- Drafted the manuscript: Hongbo HE
- Critical revision: Wenxuan HOU

This statement is signed by all the authors: Duan LIU, Zhiyuan LI, Hongbo HE and Wenxuan HOU